

ALGEBRAIC COMBINATORICS

Patricia Commins Invariant theory for the face algebra of the braid arrangement Volume 8, issue 2 (2025), p. 421-468. https://doi.org/10.5802/alco.412

© The author(s), 2025.

CREATIVE COMMONS ATTRIBUTION (CC-BY) 4.0 LICENSE. http://creativecommons.org/licenses/by/4.0/



Algebraic Combinatorics is published by The Combinatorics Consortium and is a member of the Centre Mersenne for Open Scientific Publishing www.tccpublishing.org www.centre-mersenne.org e-ISSN: 2589-5486





Invariant theory for the face algebra of the braid arrangement

Patricia Commins

ABSTRACT The faces of the braid arrangement form a monoid. The associated monoid algebra – the face algebra – is well-studied, especially in relation to card shuffling and other Markov chains. In this paper, we explore the action of the symmetric group on the face algebra from the perspective of invariant theory. Bidigare proved the invariant subalgebra of the face algebra is (anti)isomorphic to Solomon's descent algebra. We answer the more general question: what is the structure of the face algebra as a simultaneous representation of the symmetric group and Solomon's descent algebra?

Special cases of our main theorem recover the Cartan invariants of Solomon's descent algebra discovered by Garsia–Reutenauer and work of Uyemura-Reyes on certain shuffling representations. Our proof techniques involve the homology of intervals in the lattice of set partitions.

1. INTRODUCTION

In this paper, we explore the interactions between the symmetric group and two related finite dimensional algebras: Solomon's descent algebra and the face algebra of the braid arrangement.

The descent algebra is a remarkable subalgebra of the symmetric group algebra – and more generally, a Coxeter group algebra – which was discovered by Solomon in [54]. There has since been great progress in understanding its representation theory (see [6, 9, 10, 12, 16, 17, 25, 41, 50, 48, 53]). However, as a non-semisimple algebra, its structure is rich and many open questions (such as the Cartan invariants in general Coxeter type) remain. The descent algebra is also studied for its connections with the free Lie algebra [25, 44], higher Lie representations [51, 45], and characters of Coxeter groups [15, 31, 54]. See [52] for a very nice overview of these connections.

The faces of any hyperplane arrangement form a semigroup under a product first considered by Tits in [63]. This face semigroup is perhaps best known for its connections to card shuffling and other Markov chains, which were pioneered by Bidigare–Hanlon–Rockmore in [13] and continued by many others (see, for example, [21, 22, 34, 43, 45]). Beyond shuffling, hyperplane face semigroups are also interesting in their own right. For instance, the structure of the associated semigroup algebras is governed by the combinatorics of the arrangement. Face semigroup algebras have been studied extensively from this perspective by Aguiar–Mahajan in [2],

Manuscript received 11th June 2024, revised 16th December 2024, accepted 17th December 2024.

KEYWORDS. Solomon's descent algebra, higher Lie characters, plethysm, finite dimensional algebras, poset topology, reflection arrangements, Tits semigroup, left regular band.

 $[\]ensuremath{\operatorname{ACKNOWLEDGEMENTS}}$. The author was partially supported by an NSF Graduate Research Fellowship.

Saliola in [50, 48, 49], and Schocker in [53]. Finally, face semigroups are also studied as a canonical example of a left regular band, a special type of idempotent semigroup studied in [1, 20, 38, 39, 47, 57].

The face algebra of the braid arrangement and the descent algebra are intrinsically linked by the symmetric group, which acts on the face algebra by algebra automorphisms. Specifically, in his PhD thesis [14], Bidigare proved the invariant subalgebra of the face algebra is (anti)isomorphic to the descent algebra. Bidigare's connection provides a useful perspective on the descent algebra. As Schocker wrote in [52], it offers "fully transparent proofs of [several of] Solomon's results and sheds new light on the representation theory of the descent algebra." As concrete examples, Saliola used this geometric viewpoint of the descent algebra to compute its quiver in types Aand B in [48] and its Loewy length in type D in [50].

Inspired by classical invariant theory, we aim to build on this story by further exploring the connections between the descent algebra, the face algebra, and the symmetric group. In particular, when a finite group G acts on a ring R by ring automorphisms, it is natural to consider the structure of R as a module over its invariant ring R^G . Bidigare's result reveals the face algebra is a (right) module over Solomon's descent algebra, opening the door to studying a concrete, combinatorial representation of the descent algebra. In [53, §9.3], Schocker initiated study in this direction. Specifically, he studied the projective indecomposable and simple modules of the face algebra as descent algebra modules in [53, Prop 9.4, Prop 9.6, Thm 9.7].

Invariant theory also asks a more refined question. Provided R is an algebra over a field of characteristic zero, R decomposes into its *G*-isotypic subspaces,

$$R = \bigoplus_{\chi} R^{\chi}$$

as χ varies over the irreducible characters of G. The trivial isotypic subspace is precisely the invariant subalgebra of R and each isotypic subspace R^{χ} is a module over the invariant subalgebra R^G . Studying each G-isotypic subspace as a module over the invariant subalgebra examines R as a module over both G and R^G simultaneously.

It is thus a natural extension of Bidigare's result to consider the other symmetric group isotypic subspaces of the face algebra. In fact, although he did not consider its module structure over the descent algebra, Bidigare also studied the structure of the sign isotypic subspace of the face algebra in his thesis [14, §3.5.3]. In what follows, let S_n be the symmetric group on n letters and let \mathbb{CF}_n denote the face algebra (over the complex numbers \mathbb{C}) of the associated braid arrangement. Our main question is as follows.

QUESTION 1.1. What is the structure of each S_n -isotypic subspace of \mathbb{CF}_n as a module over the descent algebra?

Our answer to Question 1.1 is (informally) summarized in the following theorem, which combines Proposition 3.3, Proposition 3.5, and Theorem 5.39. The irreducible representations of both the symmetric group and the descent algebra are indexed by partitions of n. Let χ^{ν} be the irreducible character of S_n associated to the partition ν and define f^{ν} to be its dimension. Let M_{λ} be the irreducible representation of the descent algebra associated to the partition λ . We write s_{ν} for the Schur function indexed by the partition ν and L_{ρ} for the symmetric function corresponding to the higher Lie character indexed by the partition ρ . Any remaining notation is discussed in Section 4. THEOREM 1.2. The χ^{ν} -isotypic subspace of the face algebra \mathbb{CF}_n has a decomposition into descent algebra submodules N^{ν}_{μ} labelled by partitions μ of n:

$$(\mathbb{C}\mathcal{F}_n)^{\chi^{\nu}} = \bigoplus_{\substack{\nu \ weakly\\ dominates \ \mu}} N_{\mu}^{\nu}.$$

The \mathbb{C} -dimensions of the descent algebra modules N^{ν}_{μ} have simple combinatorial formulas and the composition multiplicity of the descent algebra simple M_{λ} within N^{ν}_{μ} is

 $f^{\nu} \cdot \langle s_{\nu}, coefficient \ of \mathbf{y}_{\lambda} \mathbf{z}_{\mu} \ in \ \Omega \rangle,$

where Ω is the generating function

$$\Omega = \prod_{\substack{Lyndon \\ words w}} \sum_{\substack{partitions \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \ \mathbf{z}_{w^{|\rho|}} \ L_{\rho}[h_w].$$

The composition multiplicities in the special cases $\nu = n$ and $\mu = 1^n$ recover work of Garsia–Reutenauer on the Cartan invariants of the descent algebra and Uyemura-Reyes on certain shuffling representations, respectively.

Outline of paper. In Section 2, we provide background on the descent algebra, the face algebra of the braid arrangement, Bidigare's theorem, and some representation theoretic concepts. In Section 3 we provide a preliminary answer (Proposition 3.3) which reduces Question 1.1 to a more refined question (Question 3.4). In Section 4 we state and give examples of our main theorem (Theorem 5.39), explain how it recovers work of Uyemura-Reyes and Garsia–Reutenauer, and give a more explicit answer for the sign isotypic subspace (Proposition 4.6). Section 5 is dedicated to the proof of our main theorem. For readability, the proof is first outlined in Section 5.1 and then broken into smaller subsections. Finally, we conclude with a brief discussion on generalizations to other Coxeter types and left regular bands in Section 6.

2. Background

2.1. THE BRAID ARRANGEMENT AND ITS FACE ALGEBRA. In this section, we briefly review the basics of the braid arrangement; we refer the reader to [2, Ch. 6] for a more detailed discussion. Write $\mathbf{x} \coloneqq (x_1, x_2, \ldots, x_n)$ to denote an element of the vector space \mathbb{R}^n . The **braid arrangement** \mathcal{B}_n is the hyperplane arrangement in \mathbb{R}^n consisting of the hyperplanes $\{\mathbf{x} : x_i = x_j\}$ for all $1 \le i < j \le n$. Each hyperplane $\{\mathbf{x} : x_i = x_j\}$ partitions \mathbb{R}^n into three subsets: the halfspace $H_{ij}^+ = \{\mathbf{x} : x_i > x_j\}$, the halfspace $H_{ij}^- = \{\mathbf{x} : x_i < x_j\}$, and the hyperplane itself H_{ij}^0 . The **faces** of \mathcal{B}_n are the nonempty intersections of the form

$$\bigcap_{1 \leqslant i < j \leqslant n} H_{ij}^{\operatorname{sgn}_{ij}}$$

for some set of choices sgn_{ij} in $\{+, -, 0\}$.

The faces of \mathcal{B}_n naturally correspond to strings of inequalities relating all coordinates. For example, one face F of \mathcal{B}_7 corresponds to the string $x_4 < x_1 = x_5 < x_7 < x_2 = x_3 = x_6$. Combinatorially, these strings (and their corresponding faces) are **ordered set partitions** of the set $[n] \coloneqq \{1, 2, \ldots, n\}$. For example, F corresponds to the ordered set partition ($\{4\}, \{1, 5\}, \{7\}, \{2, 3, 6\}$), which we write as (4, 15, 7, 236). The symmetric group S_n acts on the faces of \mathcal{B}_n by $w(P_1, P_2, \ldots, P_k) \coloneqq (w(P_1), w(P_2), \ldots, w(P_k))$ for $w \in S_n$. By taking block-sizes, the S_n -orbits of faces in \mathcal{F}_n are indexed by **integer compositions** of n, written $\alpha \models n$, meaning $\alpha = (\alpha_1, \alpha_2, \ldots, \alpha_k)$ is a sequence of positive integers summing to n.

EXAMPLE 2.1. The braid arrangement \mathcal{B}_3 (intersected with the plane $x_1 + x_2 + x_3 = 0$) and its thirteen faces are shown below. The table points out the four S_3 -orbits of faces indexed by the compositions $\alpha \models 3$.



The faces of \mathcal{B}_n have an associative, *noncommutative* multiplicative structure, first considered by Tits in [63]. In terms of ordered set partitions,

$$(P_1, P_2, \dots, P_k) \cdot (Q_1, Q_2, \dots, Q_\ell) \coloneqq (P_1 \cap Q_1, P_1 \cap Q_2, \dots, P_1 \cap Q_\ell, P_2 \cap Q_1, \dots, P_k \cap Q_\ell)^{\wedge},$$

where \wedge indicates the removal of empty sets. For example, in \mathcal{B}_7 ,

$$(4, 15, 7, 236) \cdot (245, 367, 1) = (4, 5, 1, 7, 2, 36).$$

The ordered set partition with a single block (12...n) is an identity element, so the faces have the structure of a monoid, which we denote by \mathcal{F}_n . Note that for any $F, G \in \mathcal{F}_n$,

(1)
$$FGF = FG.$$

By setting $G = (12 \cdots n)$ in Eq. (1), one sees that each element in \mathcal{F}_n is an idempotent. In fact, Eq. (1) implies that \mathcal{F}_n belongs to a very special class of idempotent semigroups, called left regular bands, which we (briefly) return to in Section 6.2.

Let Π_n denote the lattice of *unordered* set partitions of [n] under the refinement ordering, so that the set partition with all numbers in a single block is a maximum element. There is a map, often called the **support map**

$$\sigma: \mathcal{F}_n \to \Pi_n$$

given by removing the ordering on the blocks of the ordered set partitions. For instance $\sigma((4, 15, 7, 236))$ is the unordered set partition $\{15, 236, 4, 7\}$. This map has the special property that for faces $F, G \in \mathcal{F}_n$,

(2)
$$\sigma(FG) = \sigma(F) \wedge \sigma(G),$$

where \wedge denotes the **meet** operation of a lattice.

Symmetric group actions are of central importance in this paper. It is routine to check that the action of S_n on \mathcal{F}_n is by monoid automorphisms. We shall also consider the standard action of S_n on Π_n , defined as $w \{P_1, \ldots, P_k\} = \{w(P_1), \ldots, w(P_k)\}$ for $w \in S_n$. One can check that the support map σ commutes with this action of S_n . The S_n -orbits of set partitions in Π_n are indexed by (integer) partitions λ whose parts sum to n (which we write as $\lambda \vdash n$). For a set partition $X \in \Pi_n$, and integer partition $\lambda \vdash n$ we write $X \in \lambda$ if the blocks of X have sizes given by the parts of λ . For instance, $\{15, 236, 4, 7\} \in (3, 2, 1, 1)$.

Rather than the monoid \mathcal{F}_n itself, we are primarily interested in its monoid algebra, which we call the **face algebra** $\mathbb{C}\mathcal{F}_n$. The face algebra $\mathbb{C}\mathcal{F}_n$ consists of formal \mathbb{C} linear combinations of faces in \mathcal{F}_n ; its multiplication is the \mathbb{C} -linear extension of multiplication in \mathcal{F}_n . More formally, we multiply the element of the face algebra with coefficients $\{c_F \in \mathbb{C} : F \in \mathcal{F}_n\}$ by the element with coefficients $\{d_G \in \mathbb{C} : G \in \mathcal{F}_n\}$ as:

$$\left(\sum_{F\in\mathcal{F}_n} c_F F\right) \cdot \left(\sum_{G\in\mathcal{F}_n} d_G G\right) \coloneqq \sum_{F,G\in\mathcal{F}_n} c_F d_G FG.$$

Since the symmetric group acts on \mathcal{F}_n by monoid automorphisms, its linearly extended action on $\mathbb{C}\mathcal{F}_n$ is by algebra automorphisms. The **invariant subalgebra** of the face algebra, written as $(\mathbb{C}\mathcal{F}_n)^{S_n}$, consists of the elements $x \in \mathbb{C}\mathcal{F}_n$ for which w(x) = x for all $w \in S_n$. Observe that the action of the symmetric group algebra $\mathbb{C}S_n$ commutes with both the left- and right- actions of the invariant subalgebra on the face algebra. In particular, for $x \in \mathbb{C}S_n$, $y \in \mathbb{C}\mathcal{F}_n$, and $z \in (\mathbb{C}\mathcal{F}_n)^{S_n}$, the following hold:

(3)
$$x(yz) = x(y)z$$
, and $x(zy) = zx(y)$.

2.2. SOLOMON'S DESCENT ALGEBRA AND BIDIGARE'S THEOREM. Each permutation $w \in S_n$ has an associated (right) **descent set** $\text{Des}(w) \coloneqq \{i : w(i) > w(i+1)\} \subseteq [n-1]$. For each subset $J \subseteq [n-1]$, define an element \mathbf{x}_J in $\mathbb{C}S_n$ by

$$\mathbf{x}_{\mathbf{J}} \coloneqq \sum_{\substack{w \in S_n: \\ \mathrm{Des}(w) \subseteq J}} w.$$

In [54], Solomon proved that the \mathbb{C} - span of the elements $\{x_J : J \subseteq [n-1]\}$ is closed under multiplication, so it is a subalgebra of $\mathbb{C}S_n$. The structure constants with respect to this basis have a simple combinatorial formula (see⁽¹⁾ [25, Proposition 1.1]). This subalgebra is known as **Solomon's descent algebra**, or simply the descent algebra, which we will denote by Σ_n . The descent algebra is intimately linked to the face algebra. In particular, for a subset $J = \{a_1 < a_2 < \ldots < a_k\} \subseteq [n-1]$, write $\alpha(J)$ to be the integer composition $(a_1, a_2 - a_1, a_3 - a_2, \ldots, a_k - a_{k-1}, n - a_k)$. Bidigare proved the following connection in [14, Theorem 3.8.1].

THEOREM 2.2. (Bidigare) The S_n -invariant subalgebra of the face algebra is antiisomorphic to Solomon's descent algebra. In terms of the S_n -orbits explained just before Example 2.1, an antiisomorphism from Σ_n to $(\mathbb{CF}_n)^{S_n}$ is given by

$$\Phi: x_J \mapsto \sum_{\substack{Faces \ F\\ in \ S_n \text{-} orbit\\ indexed \ by \ \alpha(J)}} F.$$

EXAMPLE 2.3. In one-line notation, the element $x_{\{1\}} = 123 + 213 + 312 \in \Sigma_3$ is mapped under Bidigare's antiisomorphism to the sum of the rays labelled by (1, 23), (2, 13), and (3, 12) in Example 2.1.

2.3. REPRESENTATION THEORY OF FINITE DIMENSIONAL ALGEBRAS. In order to explain the representation theory of the descent algebra, we first give a brief overview of the representation theory of finite dimensional algebras. We follow the explanations given in [2, Appendix D], [5, Ch. 1], [23, Ch. 9], [39, §4.2], and [66, Ch. 6, 7].

A \mathbb{C} -algebra A, which we will often just call an algebra, is a \mathbb{C} -vector space which is also a ring with unit in which c(ab) = (ac)b = a(cb) = (ab)c for all $c \in \mathbb{C}, a, b \in A$. We will assume A is **finite dimensional**, which means it is finite dimensional as a

⁽¹⁾In the notation of [25], x_J is $B_{\alpha(J)}$, where $\alpha(J)$ is the composition we define before Theorem 2.2.

 \mathbb{C} -vector space. A left (or right) **representation of an algebra** A is a left (or right) A-module, so we will use the language of modules and representations somewhat interchangeably. For simplicity, we assume we are dealing with left⁽²⁾ A-modules for the remainder of this subsection, but the corresponding analogues hold for right A-modules too. An A-module U is called

- indecomposable if it cannot be written as a nontrivial direct sum of A-modules $U = U_1 \oplus U_2$.
- **simple** if it has no nontrivial *A*-submodules.
- **projective** if there is a free A-module F and another A-module U' such that $F \cong U \oplus U'$ as A-modules.

The **idempotents** in an algebra are the elements E for which $E^2 = E$; these play a key role in the representation theory of finite dimensional algebras. A family E_1, E_2, \ldots, E_n of nonzero idempotents in A is said to be

- complete if $E_1 + E_2 + ... + E_n = 1$.
- orthogonal if $E_i E_j = 0$ for all $i \neq j$.
- primitive if for all *i*, there do not exist two nonzero orthogonal idempotents E, E' for which $E_i = E + E'$.

Complete families of primitive, orthogonal idempotents (or cfpois) $\{E_1, E_2, \ldots, E_n\}$ of A biject with decompositions of A into indecomposable left A-modules by

$$A = \bigoplus_{i=1}^{n} AE_i,$$

see, for instance, [66, Proposition 7.2.1]. The decomposition of A into indecomposables is unique, up to re-indexing and A-module isomorphisms, by the Krull–Schmidt theorem. Since each A-module AE_i is a direct summand of A, each is projective. Further, up to isomorphism, these are the only projective indecomposable modules, so the isomorphism classes of these summands are called **the projective indecomposable modules of** A (see [5, Lemma I.5.3(b)]). So, for any two cfpois $\{E_i : 1 \leq i \leq n\}$ and $\{F_j : 1 \leq j \leq m\}$ of A, we know that m = n and there exists a reordering $\sigma \in S_n$ such that $AE_i \cong AF_{\sigma(i)}$ as A-modules.

REMARK 2.4. We will use the following two facts several times. Let A be a finite dimensional \mathbb{C} -algebra. Conjugating any cfpoi of A by a unit of A produces another cfpoi of A. Conversely, any two cfpois of A are conjugate by some unit of A. (This latter fact follows from [2, Lemma D.26 and its preceding discussion].)

The **radical** of a finite dimensional algebra A, written $\operatorname{rad}(A)$, is the intersection of all of its maximal left ideals. It turns out to be a nilpotent, two-sided ideal of A([5, Corollary I.1.4, Corollary I.2.3]). More generally, the radical of an A-module U, written $\operatorname{rad}(U)$ is the intersection of all its maximal A-submodules. One can also define it as $\operatorname{rad}(A)U$ ([5, Proposition I.3.7]). Up to isomorphism, the set of simple A-modules is

$\{AE_i/\mathrm{rad}(AE_i): 1 \leqslant i \leqslant n\}$

for any choice of cfpoi $\{E_i\}$. The projective indecomposable module AE_i is a **projective cover** for the simple module $AE_i/rad(AE_i)$ ([5, Corollary I.5.9]). Further, one has an isomorphism of the simple A-modules $AE_i/rad(AE_i) \cong AE_j/rad(AE_j)$ if and only if one has a corresponding isomorphism of their projective covers $AE_i \cong AE_j$.

 $^{^{(2)}}$ Even though the isotypic subspaces of the face algebra are *right* modules over the descent algebra, we will mainly view them as *left* modules over the invariant subalgebra of the face algebra. These approaches are equivalent by Bidigare's theorem.

In general, it is possible that $AE_i \cong AE_j$ for $i \neq j$. If we condense and write the list of isomorphism classes of simple A-modules and their corresponding projective indecomposable A-modules as $\{M_\alpha\}$ and $\{P_\alpha\}$ as α ranges in some indexing set \mathcal{A} , it turns out that as A-modules,

(4)
$$A \cong \bigoplus_{\alpha \in \mathcal{A}} P_{\alpha}^{\oplus \dim_{\mathbb{C}} M_{\alpha}},$$

see for instance [66, Theorem 7.3.9].

An algebra A is **semisimple** if every A-module can be decomposed into a direct sum of simple A-modules. By the Artin–Wedderburn theorem, an algebra A is semisimple if and only if rad(A) = 0 ([5, Theorem I.3.4]). Maschke's theorem gives that group algebras $\mathbb{C}G$ for finite groups G are semisimple. Unlike group algebras, many finite-dimensional \mathbb{C} -algebras are *not* semisimple (including both the face algebra and the descent algebra).

When an algebra A is semisimple, the "gold standard" of understanding an A-module is to decompose it into a direct sum of simple A-modules. When A is non-semisimple, an alternative method must be used; we will take the approach of counting *composition multiplicities*. A **composition series** of a finite dimensional A-module U is a sequence of A-modules

$$0 \subsetneq U_1 \subsetneq U_2 \subsetneq \ldots \subsetneq U_{n-1} \subsetneq U$$

such that each successive quotient U_i/U_{i-1} (called a **composition factor**) is a simple A-module. Although composition series of an A-module are not unique, the Jordan–Hölder theorem explains that the composition factors (up to isomorphism) and their multiplicities are. The multiplicity of a simple A-module M as a composition factor of any composition series of an A-module U is called the **composition multiplicity** of M in U, and written as [U:M].

One way to compute composition multiplicities without constructing an explicit composition series uses the projective indecomposable A-modules as projective covers for the simple A-modules. In particular, let U be a finite dimensional A-module, let M = AE/rad(AE) be a simple A-module, and let P = AE be its projective cover. Since we are working over \mathbb{C} , it turns out that (see [66, Proposition 7.4.1], specializing to \mathbb{C}),

(5)
$$[U:M] = \dim_{\mathbb{C}} \operatorname{Hom}_{A}(P,U) = \dim_{\mathbb{C}} \operatorname{Hom}_{A}(AE,U) = \dim_{\mathbb{C}} EU.$$

A case of special importance is understanding the composition factors of the projective indecomposable modules. Let $\{M_{\alpha}\}$ and $\{P_{\alpha}\}$ be the isomorphism classes of simple A-modules and their corresponding projective indecomposables, as α varies in some indexing set \mathcal{A} . The composition multiplicities $[P_{\beta} : M_{\alpha}]$ as α, β vary in \mathcal{A} are called the **Cartan invariants** of \mathcal{A} . Using the above equation,

(6)
$$[P_{\beta}: M_{\alpha}] = \dim_{\mathbb{C}} E_{\alpha} A E_{\beta}.$$

The Cartan invariants of the descent algebra are known in types A and B, and have beautiful combinatorial interpretations; we will discuss these in Section 4.4 and Section 6.

2.4. REPRESENTATION THEORY OF SOLOMON'S DESCENT ALGEBRA. The representation theory of the descent algebra involves elegant combinatorics. Its (right) representation theory was studied in great depth by Garsia and Reutenauer in [25] using its connections with the free Lie algebra. It was soon after studied in other Coxeter types by Bergeron, Bergeron, Howlett, and Taylor in [9, 10, 12]. Atkinson recovered some of Garsia–Reutenauer's results using other methods in [6]. Bidigare's theorem provided

a new perspective on the representation theory of the descent algebra, studied by Saliola in [48], Aguiar–Mahajan in [2], and Schocker in [53].

Both the descent algebra and the face algebra turn out to belong to a well-behaved class of algebras called **elementary algebras** (meaning A/rad(A) is a commutative algebra isomorphic to \mathbb{C}^n for some n). A \mathbb{C} -algebra is elementary precisely if its simple modules are all one-dimensional. Aguiar–Mahajan have a very nice appendix [2, Appendix D.8] explaining properties of elementary algebras to which we will sometimes refer.

2.4.1. Radical. Recall that the descent algebra Σ_n has dimension 2^{n-1} . In Solomon's original paper [54], he proved the radical $\operatorname{rad}(\Sigma_n)$ has dimension $2^{n-1} - \#\{\lambda \vdash n\}$ and is spanned by the differences $x_J - x_{J'}$ for which $\alpha(J)$ and $\alpha(J')$ (defined just before Theorem 2.2) rearrange to the same partition. Hence, Σ_n is not semisimple for all $n \ge 3$.

2.4.2. Simples and projective indecomposables. As we will see in Section 2.4.3, the idempotents in a cfpoi for the descent algebra Σ_n are indexed by integer partitions of n. Since Σ_n is elementary, each simple Σ_n -module is one-dimensional. By Eq. (4), each idempotent in a cfpoi for Σ_n then generates a distinct (up to isomorphism) projective indecomposable module, so the Σ_n - simples and projective indecomposables are also indexed by partitions. (See also [2, Theorem D.35].) One can see an explicit construction of these simples in [6, §3]. We will write M_{λ} to denote the Σ_n -simple indexed by the partition λ and P_{λ} for the projective indecomposable which is the projective cover of M_{λ} . Then,

$$\Sigma_n = \bigoplus_{\lambda \vdash n} P_{\lambda}.$$

2.4.3. Idempotents. Cfpois for the descent algebras are well-studied. Initially, they were studied from the perspective of Coxeter groups (see [9, 10, 12, 25]). Using Bidigare's theorem, Saliola provided a geometric perspective by constructing a cfpoi [48] from the cfpois he discovered for the face algebra [48, 49]. Aguiar–Mahajan further study Saliola's idempotents and generalizations extensively in [2]. For work connecting both perspectives, see [18] and [2, \$16]. For our purposes, the geometric perspective shall prove very useful, so we shall consider cfpois $\{E_{\lambda} : \lambda \vdash n\}$ of $(\mathbb{C}\mathcal{F}_n)^{S_n}$, rather than the (equivalent) cfpois $\{\Phi^{-1}(E_{\lambda}): \lambda \vdash n\}$ of Σ_n , where Φ is Bidigare's antiisomorphism from Theorem 2.2.

The idempotents in a cfpoi of the face algebra $\mathbb{C}\mathcal{F}_n$ turn out to be indexed by the set partitions in Π_n . In [48, §5], Saliola constructed explicit cfpois⁽³⁾ { $\hat{E}_X : X \in \Pi_n$ } for $\mathbb{C}\mathcal{F}_n$ for which $w\left(\widehat{E}_X\right) = \widehat{E}_{w(X)}$ for all $w \in S_n$. He proved the S_n -orbit sums of such families $\{\widehat{E}_{\lambda} : \lambda \vdash n\}$ where $\widehat{E}_{\lambda} := \sum_{X \in \lambda} \widehat{E}_X$, form cfpois for $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Thus, their inverse images under Bidigare's antiisomorphism form a cfpoi for Σ_n .

Although we will not need Saliola's explicit construction for our purposes, we will repeatedly use that his idempotents \widehat{E}_X satisfy three very nice properties⁽⁴⁾:

- (a) For $F \in \mathcal{F}_n$, if $\sigma(F) \not\geq X$, then $F \widehat{E}_X = 0$. ([48, Lemma 5.3])
- (b) $\widehat{E}_X \in \operatorname{span}_{\mathbb{C}} \{F \in \mathcal{F}_n : \sigma(F) \leq X\}$. ([48, Theorem 5.2]) (c) Let $\mathfrak{c}_X : \mathbb{C}\mathcal{F}_n \to \mathbb{C}$ be the \mathbb{C} -linear map which sends $F \in \mathcal{F}_n$ to 1 if $\sigma(F) = X$ and sends F to 0 otherwise. Then, $\mathfrak{c}_X(\widehat{E}_X) = 1$ ([48, Theorem 5.2]).

⁽³⁾Saliola's construction was quite general and provided multiple possible explicit cfpois (see [48, §5.1, 5.2] for some specific examples). When we write $\{E_X\}$, we mean any of the cfpois for \mathbb{CF}_n coming from his construction in [48, Theorem 5.2].

⁽⁴⁾Notation warning: Saliola uses the opposite ordering as this paper on the poset Π_n .

By conjugating Saliola's idempotents, properties (a) and (b) turn out to hold more generally for any cfpoi of $\mathbb{C}\mathcal{F}_n$. We should point out that each of the lemmas in the remainder of this idempotent section appear (often with much stronger statements and in more generality) in the remarkable characterizations of cfpois for $\mathbb{C}\mathcal{F}_n$ and $(\mathbb{C}\mathcal{F}_n)^{S_n}$ given by Aguiar–Mahajan [2, Chapters 11, 16, Appendix D]. For simplicity, we state only the minimal information that we will need, but we highly encourage the curious reader to consult their work.

LEMMA 2.5. Let $\{E_X : X \in \Pi_n\}$ be $ang^{(5)}$ cfpoi for $\mathbb{C}\mathcal{F}_n$. The following statements hold.

- (i) Let $F \in \mathcal{F}_n$. If $\sigma(F) \not\geq X$, then $FE_X = 0$.
- (ii) $E_X \in \operatorname{span}_{\mathbb{C}} \{ F \in \mathcal{F}_n : \sigma(F) \leq X \}.$
- (iii) If $Y \not\geq X$, then $E_Y \mathbb{C} \mathcal{F}_n E_X = 0$.
- (iv) Let $F \in \mathcal{F}_n$. If $\sigma(F) = X$, then there are constants $d_{F'} \in \mathbb{C}$ such that FE_X can be written as

$$FE_X = F + \sum_{F':\sigma(F') < X} d_{F'} F'$$

Proof. Recall $\{E_X\}$ is conjugate to the family $\{\widehat{E}_X\}$ by some invertible element of $\mathbb{C}\mathcal{F}_n$, say $u := \sum_{F \in \mathcal{F}_n} c_F F$.

(i) Let $F \in \mathcal{F}_n$ with $\sigma(F) \not\geq X$. By Eq. (2), for any $G \in \mathcal{F}_n, \sigma(FG) = \sigma(F) \land \sigma(G) \not\geq X$. Using property (a) of Saliola's idempotents,

$$FE_X = Fu\widehat{E}_X u^{-1} = \left(\sum_{G \in \mathcal{F}_n} c_G \ (FG)\widehat{E}_X\right) u^{-1} = 0.$$

- (ii) It follows from Eq. (2) that $\operatorname{span}_{\mathbb{C}}\{F \in \mathcal{F}_n : \sigma(F) \leq X\}$ is closed under left and right multiplication by $\mathbb{C}\mathcal{F}_n$. Property (b) of Saliola's idempotents explains that \widehat{E}_X is in $\operatorname{span}_{\mathbb{C}}\{F \in \mathcal{F}_n : \sigma(F) \leq X\}$. Hence, $\widehat{E}_X = uE_Xu^{-1}$ is also in $\operatorname{span}_{\mathbb{C}}\{F \in \mathcal{F}_n : \sigma(F) \leq X\}$.
- (iii) By part (ii), $E_Y \in \operatorname{span}_{\mathbb{C}} \{F \in \mathcal{F}_n : \sigma(F) \leq Y\}$. For F with $\sigma(F) \leq Y$, $\sigma(F) \not\geq X$ (since otherwise, $Y \geq \sigma(F) \geq X$). By part (i), $E_Y \mathbb{C} \mathcal{F}_n E_X = 0$.
- (iv) By part (ii), it suffices to prove that the coefficient of F in FE_X is one and the coefficient of $F' \neq F$ for $\sigma(F') = X$ is zero. By property (b) of Saliola's idempotents, there exist constants $d_{F'}$ for which

$$FE_X = Fu\widehat{E}_X u^{-1} = Fu\left(\sum_{\substack{F' \in \mathcal{F}_n:\\\sigma(F') \leqslant X}} d_{F'}F'\right) u^{-1} = \sum_{\substack{F' \in \mathcal{F}_n:\\\sigma(F') \leqslant X}} d_{F'} FuF' u^{-1}.$$

By expanding u, u^{-1} as a linear combination of faces and applying Eq. (2), observe that if $\sigma(F') < X$, then $\mathfrak{c}_X(FuF'u^{-1}) = 0$. Hence, we can ignore the terms with $\sigma(F') < X$ the above sum. From the ordered set partition definition of face multiplication, observe that if $\sigma(F') = X$, then FF' = F. Using this fact and Eq. (1),

$$FuF'u^{-1} = FuFF'u^{-1} = FuFu^{-1} = Fuu^{-1} = F.$$

⁽⁵⁾We assume appropriate indexing here, in that if u is an invertible element of \mathbb{CF}_n which conjugates $\{\widehat{E}_X : X \in \Pi_n\}$ to produce $\{E_X : X \in \Pi_n\}$, then $E_X = u\widehat{E}_X u^{-1}$ for all $X \in \Pi_n$. Put differently, for all $X \in \Pi_n$, E_X and \widehat{E}_X lift the same idempotent in the unique cfpoi of $\mathbb{CF}_n/\mathrm{rad}(\mathbb{CF}_n)$. (See [2, Thm D.33].)

Algebraic Combinatorics, Vol. 8 #2 (2025)

Hence, the coefficient of $F' \neq F$ with $\sigma(F') = X$ in FE_X is 0 as desired, and the coefficient of F in FE_X is $\sum_{F':\sigma(F')=X} d_{F'} = 1$ by property (c) of Saliola's idempotents.

The proof of the following lemma imitates the proofs of Lemma 11.29 in [2] and Corollary 4.4 of [39].

LEMMA 2.6. Let $\{E_X : X \in \Pi_n\}$ be any cfpoi for $\mathbb{C}\mathcal{F}_n$. The set $\{FE_{\sigma(F)} : F \in \mathcal{F}_n\}$ forms a \mathbb{C} -basis for the vector space $\mathbb{C}\mathcal{F}_n$.

Proof. Pick a total ordering on the faces $F \in \mathcal{F}_n$ such that if $\sigma(F) < \sigma(F')$, then F precedes F'. Then, the matrix representing the linear transformation $F \mapsto FE_{\sigma(F)}$ is upper triangular with ones along the diagonal by Lemma 2.5(iv). Hence, $\{FE_{\sigma(f)} : F \in \mathcal{F}_n\}$ is also a basis for $\mathbb{C}\mathcal{F}_n$.

LEMMA 2.7. Let $\{E_{\lambda} : \lambda \vdash n\}$ be any cfpoi for $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Then, there exists a cfpoi $\{E_X : X \in \Pi_n\}$ for the face algebra $\mathbb{C}\mathcal{F}_n$ which satisfies all three of the following properties:

- (i) $w(E_X) = E_{w(X)}$ for all $X \in \Pi_n$ and all $w \in S_n$,
- (ii) $E_{\lambda} = \sum_{X \in \lambda} E_X$, and
- (iii) There is an invertible $u \in (\mathbb{C}\mathcal{F}_n)^{S_n}$ such that $E_X = u\widehat{E}_X u^{-1}$ for all $X \in \Pi_n$.

Proof. Let $\{E_{\lambda} : \lambda \vdash n\}$ be a cfpoi for $(\mathbb{C}\mathcal{F}_n)^{S_n}$. By Remark 2.4, there exists an invertible element $u \in (\mathbb{C}\mathcal{F}_n)^{S_n}$ such that for each $\lambda \vdash n$, $E_{\lambda} = u\widehat{E}_{\lambda}u^{-1}$. Hence, for each $\lambda \vdash n$,

$$E_{\lambda} = \sum_{X \in \lambda} u \widehat{E}_X u^{-1}$$

By Remark 2.4, the family $\left\{ u \widehat{E}_X u^{-1} : X \in \Pi_n \right\}$ is a cfpoi for $\mathbb{C}\mathcal{F}_n$. Since for $w \in S_n$, $w(u^{-1}) = (w(u))^{-1} = u^{-1}$, the inverse u^{-1} is also in $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Hence, for $w \in S_n$,

$$w\left(u\widehat{E}_Xu^{-1}\right) = w(u)w\left(\widehat{E}_X\right)w(u^{-1}) = u\widehat{E}_{w(X)}u^{-1}.$$

Setting $E_X \coloneqq u \widehat{E}_X u^{-1}$ completes the proof.

A similar statement to Lemma 2.5(ii) holds for the cfpois of the invariant subalgebra $(\mathbb{C}\mathcal{F}_n)^{S_n}$.

LEMMA 2.8. Let $\{E_{\mu} : \mu \vdash n\}$ be any choice of cfpoi for $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Let $F \in \mathcal{F}_n$ with $\sigma(F) \in \lambda$. If the partition μ does not refine λ , then

$$FE_{\mu} = 0.$$

Proof. By Lemma 2.7, there exists a cfpoi $\{E_X : X \in \Pi_n\}$ for \mathbb{CF}_n permuted by S_n for which $E_\mu = \sum_{X \in \mu} E_X$. If $\sigma(F) \ge X$ for some $X \in \mu$, then the set partition X refines $\sigma(F)$, so μ would refine λ . Hence, $\sigma(F) \ge X$ for all $X \in \mu$. Thus, by Lemma 2.5(ii),

$$FE_{\mu} = \sum_{X \in \mu} FE_X = 0.$$

For the remainder of the paper, we shall fix:

- some cfpoi $\{E_{\lambda} : \lambda \vdash n\}$ of $(\mathbb{C}\mathcal{F}_n)^{S_n}$, and
- a cfpoi $\{E_X : X \in \Pi_n\}$ for $\mathbb{C}\mathcal{F}_n$ satisfying the three conditions of Lemma 2.7 (given our choice of $\{E_\lambda\}$).

2.5. Representation theory of the symmetric group and wreath prod-UCTS. We will assume some familiarity with the representation theory of finite groups (especially symmetric groups) as well as symmetric functions. See [46], [56, Ch. 7], and [4] for good references.

Let χ^{λ} denote the irreducible character of S_n indexed by partition λ ; we will often write $\mathbb{1}_{S_n}$ and sgn_{S_n} to denote the trivial and sign characters of S_n , respectively. Throughout this paper, let ch denote the Frobenius characteristic map from characters of symmetric groups (or their corresponding representations) to the ring of symmetric functions. We will write s_{λ} for the Schur function indexed by λ and h_i for the complete homogeneous symmetric function of degree *i*. Recall that

- ch (χ^λ) = s_λ,
 ch (sgn_{S_n}) = ch (χ^{1ⁿ}) = s_{1ⁿ}, and
 ch (𝔅_n) = ch (χⁿ) = s_n = h_n.

We use the convention $h_{\alpha} \coloneqq h_{\alpha_1} h_{\alpha_2} \dots h_{\alpha_k}$ for α an integer composition, integer partition, or even a word on the positive integers. For α and α' integer compositions, integer partitions, or words on the positive integers, we write $\alpha \sim \alpha'$ if α rearranges to α' . Note that $h_{\alpha} = h_{\alpha'}$ if $\alpha \sim \alpha'$. For $\alpha \models n$, we shall view $S_{\alpha} \coloneqq S_{\alpha_1} \times S_{\alpha_2} \times \ldots \times S_{\alpha_k}$ as a subgroup of S_n , by considering it as the subgroup which permutes the first α_1 positive integers amongst themselves, the next α_2 amongst themselves, and so on.

Recall that for $\rho_1, \rho_2, \ldots, \rho_k$ characters of $S_{\alpha_1}, S_{\alpha_2}, \ldots, S_{\alpha_k}$,

$$\operatorname{ch}\left(\left(\rho_1\otimes\rho_2\otimes\ldots\otimes\rho_k\right)\Big|_{S_{\alpha}}^{S_n}\right)=\operatorname{ch}\left(\rho_1\right)\operatorname{ch}\left(\rho_2\right)\ldots\operatorname{ch}\left(\rho_k\right),$$

where the notation $(-)\uparrow_{H}^{G}$ indicates the **induction** of a representation of a subgroup H to the full group G. Later on, we will also consider the **restriction** of a G-representation to an H-representation, written $(-) \downarrow_H^G$.

We use the inner product symbol $\langle A, B \rangle$ to mean either

- the Hall inner product on symmetric functions if A and B are symmetric functions, or
- The inner product on the characters of a group G if A and B are Gcharacters (or the inner product on the corresponding characters if A, B are G-representations).

Recall that for S_n -representations ρ, ψ , one has $\langle \rho, \psi \rangle = \langle ch(\rho), ch(\psi) \rangle$.

We remind the reader of **Young's rule** (see [46, §2.11]). Let λ, μ be partitions of *n*. Recall the Kostka coefficients $K_{\lambda,\mu}$ which count semistandard Young tableaux with shape λ and content μ . Young's rule states that

(7)
$$\left\langle \mathbb{1}\uparrow_{S_{\mu}}^{S_{n}},\chi^{\lambda}\right\rangle = \langle h_{\mu},s_{\lambda}\rangle = K_{\lambda,\mu}.$$

It is well known that $K_{\lambda,\mu} = 0$ if λ does not (weakly) dominate μ .

Wreath products of symmetric groups with finite groups will play an important role in this work. Let G be a finite group. The wreath product $S_n[G]$ is defined to be the group with elements $\{(g_1, g_2, \dots, g_n; \sigma) : g_i \in G, \sigma \in S_n\}$ under the operation

$$(g_1, g_2, \dots, g_n; \sigma) \cdot (h_1, h_2, \dots, h_n; \tau) = (g_1 h_{\sigma^{-1}(1)}, g_2 h_{\sigma^{-1}(2)}, \dots, g_n h_{\sigma^{-1}(n)}; \sigma\tau)$$

Let V be a representation of S_n and let W be a representation of G. We write V[W], like Wachs [65] and Sundaram [59], to denote the $S_n[G]$ representation with underlying vector space

$$(\otimes^n W) \otimes V,$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

and action

$$(g_1, g_2, \dots, g_n; \sigma) \cdot (w_1 \otimes w_2 \otimes \dots \otimes w_n) \otimes v$$

= $(g_1 \cdot w_{\sigma^{-1}(1)} \otimes g_2 \cdot w_{\sigma^{-1}(2)} \otimes \dots \otimes g_n \cdot w_{\sigma^{-1}(n)}) \otimes \sigma \cdot v.$

We shall often consider the wreath product $S_m[S_n]$ as a subgroup of S_{mn} , as illustrated by the following example.

EXAMPLE 2.9. We consider $S_3[S_4]$ as the subgroup of S_{12} generated by the following permutations (in cycle notation):

• the nine transpositions

(1, 2), (2, 3), (3, 4), (5, 6), (6, 7), (7, 8), (9, 10), (10, 11), (11, 12)

which generate the subgroup S_{4^3} and are represented by the nine horizontal arrows *within* the rows of Fig. 1, and

• the two permutations

 $r_1 = (1,5)(2,6)(3,7)(4,8)$ and $r_2 = (5,9)(6,10)(7,11)(8,12)$

which swap *entire rows* in Fig. 1, represented by the two labelled vertical arrows.



FIGURE 1. $S_3[S_4] \leq S_{12}$

Wreath products are highly related to the notion of **plethysm** f[g] of symmetric functions f, g. (We assume familiarity with plethysm and recommend [56, Ch. 7 Appendix 2], [35], [60, §4] and [37, §1.8] as references.) If V is a representation of S_m with ch(V) = f and W a representation of S_n with ch(W) = g, then

$$\operatorname{ch}\left(V[W]\big\uparrow_{S_m[S_n]}^{S_{mn}}\right) = f[g].$$

The following properties of plethysm hold for general symmetric functions: see Equation (8.7) and Remark (8.3) of [37].

LEMMA 2.10. Let f, g, and h be symmetric functions. Then,

- (1) (Associativity of plethysm): f[g[h]] = f[g][h].
- (2) f[h] + g[h] = (f+g)[h]
- (3) $(f[h]) \cdot (g[h]) = (fg)[h]$

It remains an open problem in algebraic combinatorics to understand plethysm coefficients, i.e. to understand how f[g] positively expands into Schur functions for two homogeneous Schur-positive symmetric functions f, g.

3. Refining the question

By Maschke's theorem, the face algebra $\mathbb{C}\mathcal{F}_n$ decomposes into a direct sum of irreducible S_n -representations. Although this decomposition is not unique, the coarser decomposition given by grouping the irreducibles of the same isomorphism type is; this is the decomposition of $\mathbb{C}\mathcal{F}_n$ into its **isotypic subspaces**. Hence, there is an S_n -representation decomposition

$$\mathbb{C}\mathcal{F}_n = \bigoplus_{\nu \vdash n} \left(\mathbb{C}\mathcal{F}_n\right)^{\nu},$$

where $(\mathbb{C}\mathcal{F}_n)^{\nu}$ is the S_n -isotypic subspace associated to χ^{ν} . (In general, for a finite group G, an irreducible G-character χ , and a representation V of G, we write $(V)^{\chi}$ to denote the χ -isotypic subspace of V.) Each isotypic subspace $(\mathbb{C}\mathcal{F}_n)^{\nu}$ can be accessed by acting on the left of $\mathbb{C}\mathcal{F}_n$ by the corresponding isotypic projector $\pi_{\nu} \in \mathbb{C}S_n$, defined as

$$\pi_{\nu} = \frac{\chi^{\nu}(1)}{n!} \sum_{w \in S_n} \chi^{\nu}(w)w.$$

We will use that an element $x \in \mathbb{CF}_n$ is in the isotypic subspace $(\mathbb{CF}_n)^{\nu}$ if and only if $\pi_{\nu}(x) = x$.

The trivial isotypic subspace $(\mathbb{C}\mathcal{F}_n)^n$ of the face algebra is precisely its invariant subalgebra $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Moreover, the isotypic subspaces are not only S_n -representations; each carries an additional, rich structure as a left module over $(\mathbb{C}\mathcal{F}_n)^{S_n}$. Hence, by Bidigare's theorem, each isotypic subspace is actually a (right) module over the descent algebra Σ_n by the action

$$y \cdot x \coloneqq \Phi(x)y$$
 for $y \in (\mathbb{C}\mathcal{F}_n)^{\nu}$, $x \in \Sigma_n$.

Since Σ_n is not semisimple, we are unable in general to decompose each Σ_n -module $(\mathbb{C}\mathcal{F}_n E_\mu)^\nu$ into a direct sum of simple Σ_n -modules. As discussed in Section 2.3, we take the alternative approach of understanding composition multiplicities and we restate our main question from the introduction (Question 1.1) more explicitly.

QUESTION 3.1. What are the Σ_n -module composition multiplicities of each S_n -isotypic subspace $(\mathbb{C}\mathcal{F}_n)^{\nu}$?

We will answer Question 3.1 with Theorem 5.39. Specifically, we will reduce Question 3.1 to understanding specific symmetric group representations, whose structure we analyze up to longstanding open problems.

3.1. PRELIMINARY ANSWER. As a first step towards answering Question 3.1, we decompose each S_n -isotypic subspace $(\mathbb{C}\mathcal{F}_n)^{\nu}$ into a direct sum of smaller Σ_n -modules. Recall that we write $\alpha \vDash n$ to denote that α is an integer composition of n, and write $\alpha \sim \alpha'$ if the parts of α can be reordered to match those of α' (for α' an integer composition, partition, or word on the positive integers). We fix an integer partition μ of n.

LEMMA 3.2. The \mathbb{C} - linear map

$$\varphi: \operatorname{span}_{\mathbb{C}} \{ F \in \mathcal{F}_n : \sigma(F) \in \mu \} \longrightarrow \mathbb{C} \mathcal{F}_n E_\mu$$
$$F \mapsto F E_{\sigma(F)}$$

is an isomorphism of S_n -representations. Hence,

$$\operatorname{ch}\left(\mathbb{C}\mathcal{F}_{n}E_{\mu}\right) = \#\{\alpha \vDash n : \alpha \sim \mu\} \cdot h_{\mu}.$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

Proof. Since $E_{\mu} = \sum_{X \in \mu} E_X$, there is a vector space decomposition,

$$\mathbb{C}\mathcal{F}_n E_\mu = \bigoplus_{X \in \mu} \mathbb{C}\mathcal{F}_n E_X$$

Hence if $\sigma(F) \in \mu$, then $FE_{\sigma(F)} \in \mathbb{C}\mathcal{F}_n E_{\mu}$. The map φ is injective because

$$\varphi\left(\sum_{G\in\mathcal{F}_n}c_G\ G\right)=0\implies\sum_{G\in\mathcal{F}_n}c_G\ GE_{\sigma(G)}=0\implies c_G=0\text{ for all }G\in\mathcal{F}_n$$

by Lemma 2.6. To show φ is surjective, it suffices to prove that $\{FE_{\sigma(F)} : \sigma(F) = X\}$ spans $\mathbb{C}\mathcal{F}_n E_X$. Let $yE_X \in \mathbb{C}\mathcal{F}_n E_X$. By Lemma 2.6, there are constants c_G for which $y = \sum_{G \in \mathcal{F}_n} c_G \ GE_{\sigma(G)}$. By orthogonality,

$$yE_X = \sum_{G \in \mathcal{F}_n} c_G \ GE_{\sigma(g)}E_X = \sum_{\substack{G \in \mathcal{F}_n:\\\sigma(G) = X}} c_G \ GE_{\sigma(G)},$$

as desired. Therefore, φ is a vector space isomorphism.

The map φ respects the action of S_n since for $F \in \mathcal{F}_n$ with $\sigma(F) \in \mu$ and $w \in S_n$, $\varphi(w(F)) = w(F)E_{\sigma(wF)} = w(F)E_{w\sigma(F)} = w(F)w\left(E_{\sigma(F)}\right) = w\left(FE_{\sigma(F)}\right) = w\left(\varphi(F)\right)$.

Recall from Example 2.1 that the S_n -orbits of faces in \mathcal{F}_n are indexed by integer compositions $\alpha \models n$. The face-orbit indexed by the composition α belongs to the support-orbit labelled by the partition μ if and only if $\alpha \sim \mu$. The S_n -stabilizer of a face in the S_n -orbit labelled by α is conjugate to the subgroup S_α . Hence, using that φ is a S_n -representation isomorphism and basic properties of induced representations (see, for instance [66, Proposition 4.3.2]),

$$\operatorname{ch}\left(\mathbb{C}\mathcal{F}_{n}E_{\mu}\right) = \operatorname{ch}\left(\bigoplus_{\substack{\alpha\vDash n:\\\alpha\sim\mu}}\mathbb{1}\uparrow_{S_{\alpha}}^{S_{n}}\right) = \sum_{\substack{\alpha\vDash n:\\\alpha\sim\mu}}h_{\alpha} = \#\{\alpha\vDash n:\alpha\sim\mu\}\cdot h_{\mu}.$$

We are now in the position to give a preliminary answer to our question. In particular, the following proposition explains the modules N^{ν}_{μ} from the informal statement of the main theorem in the introduction (Theorem 1.2) as $N^{\nu}_{\mu} = (\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu}$. It also provides the promised "simple combinatorial formulas" for their dimensions over \mathbb{C} . Recall that f^{ν} denotes the dimension of χ^{ν} (or equivalently the number of standard Young tableaux of shape ν).

PROPOSITION 3.3. As (right) Σ_n -modules,

$$(\mathbb{C}\mathcal{F}_n)^{\nu} = \bigoplus_{\mu \vdash n} (\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu}.$$

The \mathbb{C} -dimension of each Σ_n -submodule $(\mathbb{C}\mathcal{F}_n E_\mu)^{\nu}$ is

$$f^{\nu} \cdot \#\{\alpha \vDash n \mid \alpha \sim \mu\} \cdot K_{\nu,\mu}.$$

In particular, if ν does not (weakly) dominate μ , then $(\mathbb{C}\mathcal{F}_n E_\mu)^\nu = 0$.

Proof. Since the idempotents $\{E_{\mu} : \mu \vdash n\} \in (\mathbb{C}\mathcal{F}_n)^{S_n}$ are complete and orthogonal, there is decomposition of Σ_n - representations,

$$\left(\mathbb{C}\mathcal{F}_{n}\right)^{\nu} = \bigoplus_{\mu \vdash n} \left(\mathbb{C}\mathcal{F}_{n}\right)^{\nu} E_{\mu}$$

By Eq. (3), $\pi_{\nu}(yE_{\mu}) = \pi_{\nu}(y)E_{\mu}$ for all $y \in \mathbb{CF}_n$. Hence, $(\mathbb{CF}_nE_{\mu})^{\nu} = (\mathbb{CF}_n)^{\nu}E_{\mu}$

Algebraic Combinatorics, Vol. 8 #2 (2025)

and there is a decomposition of Σ_n -representations

$$(\mathbb{C}\mathcal{F}_n)^{\nu} = \bigoplus_{\mu \vdash n} (\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu}.$$

By Lemma 3.2, the \mathbb{C} -dimension of $(\mathbb{C}\mathcal{F}_n E_\mu)^\nu$ is dim $(\chi^\nu) \cdot \#\{\alpha \sim \mu : \alpha \models n\} \cdot \langle s_\nu, h_\mu \rangle$. Recall from Eq. (7) that $\langle s_\nu, h_\mu \rangle = K_{\nu,\mu}$, which is 0 unless ν (weakly) dominates μ .

Proposition 3.3 reduces Question 3.1 to understanding the composition multiplicities of each Σ_n -module $(\mathbb{C}\mathcal{F}_n E_\mu)^{\nu}$ for any two partitions μ, ν of n. Thus, we have refined Question 3.1 as follows.

QUESTION 3.4. For any triple of partitions μ, ν, λ of n, what is the composition multiplicity

$$[(\mathbb{C}\mathcal{F}_n E_\mu)^\nu : M_\lambda]?$$

We can use the finite dimensional algebra tools discussed in Section 2.3 to further reduce our question.

PROPOSITION 3.5. The composition multiplicity of the simple Σ_n -module M_{λ} in $(\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu}$ is

$$\left[(\mathbb{C}\mathcal{F}_n E_\mu)^\nu : M_\lambda \right] = f^\nu \cdot \langle s_\nu, \operatorname{ch} \left(E_\lambda \mathbb{C}\mathcal{F}_n E_\mu \right) \rangle.$$

Proof. By Eq. (5) (adjusted for right modules),

$$\left[(\mathbb{C}\mathcal{F}_n E_\mu)^\nu : M_\lambda \right] = \dim_{\mathbb{C}} \left((\mathbb{C}\mathcal{F}_n E_\mu)^\nu \Phi^{-1}(E_\lambda) \right) = \dim_{\mathbb{C}} E_\lambda \left(\mathbb{C}\mathcal{F}_n E_\mu \right)^\nu.$$

By Eq. (3), $E_{\lambda}\pi_{\nu}(yE_{\mu}) = \pi_{\nu}(E_{\lambda}yE_{\mu})$ for all $y \in \mathbb{C}\mathcal{F}_n$. Hence,

$$E_{\lambda} \left(\mathbb{C} \mathcal{F}_n E_{\mu} \right)^{\nu} = \left(E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu} \right)^{\nu}$$

Therefore,

$$[(\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu} : M_{\lambda}] = \dim_{\mathbb{C}} (E_{\lambda}\mathbb{C}\mathcal{F}_n E_{\mu})^{\nu}$$
$$= \dim_{\mathbb{C}} (\chi^{\nu}) \cdot \langle \chi^{\nu}, E_{\lambda}\mathbb{C}\mathcal{F}_n E_{\mu} \rangle$$
$$= f^{\nu} \cdot \langle s_{\nu}, \operatorname{ch} (E_{\lambda}\mathbb{C}\mathcal{F}_n E_{\mu}) \rangle.$$

Given Proposition 3.5, our final conversion of Question 3.1 is the question below. QUESTION 3.6. What is the Frobenius characteristic of $E_{\lambda} \mathbb{CF}_n E_{\mu}$?

4. Our Answer

4.1. HIGHER LIE CHARACTERS AND LYNDON WORDS. Thrall studied a collection of S_n -representations in [62] which are (also) indexed by partitions of n and often called the **higher Lie representations**. We write L_{λ} to denote the Frobenius image of the higher Lie representation \mathcal{L}_{λ} associated to λ . These representations have several interesting interpretations and are closely tied to the free Lie algebra. For our purposes, it is most revealing to define⁽⁶⁾ L_n as the Frobenius image of the S_n -representation carried by the top homology of the order complex of the proper part of the set partition lattice⁽⁷⁾ tensored with the sign representation:

$$L_n \coloneqq \operatorname{ch}\left(H_{n-3}\left(\Pi_n\right) \otimes \operatorname{sgn}_n\right).$$

⁽⁶⁾This is equivalent to the standard definition by work of Stanley [55, Theorem 7.3], Hanlon [29, Theorem 4], and Klyachko [33]; see also a combinatorial proof by Barcelo in [7].

⁽⁷⁾See Section 5.2.4 for details and our conventions on poset topology.

More generally, for a partition $\lambda = 1^{m_1} 2^{m_2} \dots k^{m_k}$, let

(8)
$$L_{\lambda} \coloneqq \prod_{i=1}^{k} h_{m_i}[L_i].$$

Positively expanding L_{λ} into Schur functions is a longstanding open problem known as **Thrall's problem**. The furthest result in this direction is due to Schocker, who wrote L_{λ} as a signed sum of Schur functions [51]. One fact we will use several times is that the higher Lie representations decompose the regular representation of S_n , which implies

(9)
$$\sum_{\lambda \vdash n} L_{\lambda} = h_{1^n}$$

A Lyndon word is a finite nonempty word on alphabet $\{1, 2, ...\}$ that is lexicographically *strictly* smaller than all of its cyclic rearrangements. It is well-known (see [36, Theorem 5.1.5]) that any word on alphabet $\{1, 2, ...\}$ factors uniquely into the concatenation of weakly (lexicographically) decreasing Lyndon words. For instance,

2431122231121 = (243)(112223)(112)(1).

Finally, we must define some notation in order to state our main theorem. Let $\alpha = (\alpha_1, \alpha_2, \ldots, \alpha_k)$ be an integer composition, partition, or word on $\{1, 2, \ldots\}$. Let $\mathbf{y} = \{y_1, y_2, \ldots\}$ be an infinite variable set and m an integer. Define $\mathbf{y}_{\alpha}, |\alpha|, \alpha \cdot m, \alpha^m$ as follows.

•
$$\mathbf{y}_{\alpha} \coloneqq y_{\alpha_{1}}y_{\alpha_{2}}\dots y_{\alpha_{k}}.$$

• $|\alpha| \coloneqq \alpha_{1} + \alpha_{2} + \dots + \alpha_{k}.$
• $\alpha \cdot m \coloneqq (\alpha_{1} \cdot m, \alpha_{2} \cdot m, \dots, \alpha_{k} \cdot m)$
• $\alpha^{m} \coloneqq (\alpha_{1}, \alpha_{2}, \dots, \alpha_{k}, \alpha_{1}, \alpha_{2}, \dots, \alpha_{k}, \dots, \alpha_{1}, \alpha_{2}, \dots, \alpha_{k})$
mk total parts

4.2. THE MAIN THEOREM. We now have the necessary definitions to state our main theorem that was paraphrased as part of Theorem 1.2 in the introduction. Note that we use the convention that there exists an empty partition ().

THEOREM 5.39. There is an equality of generating functions

(10)
$$\sum_{n \ge 0} \sum_{\substack{\lambda \vdash n \\ \mu \vdash n}} \mathbf{y}_{\lambda} \mathbf{z}_{\mu} \cdot \operatorname{ch}(E_{\lambda} \mathbb{C} \mathcal{F}_{n} E_{\mu}) = \prod_{\substack{Lyndon \\ w}} \sum_{\substack{partition \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_{w}].$$

Theorem 5.39 explains the structure of \mathbb{CF}_n as a module over S_n and Σ_n simultaneously, answering Question 3.1. Indeed, Proposition 3.5 and Theorem 5.39 combine to give

(11)
$$[(\mathbb{C}\mathcal{F}_{n}E_{\mu})^{\nu}:M_{\lambda}] = f^{\nu} \cdot \left\langle s_{\nu}, \ [\mathbf{y}_{\lambda}\mathbf{z}_{\mu}] \prod_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{\text{partition} \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_{w}] \right\rangle,$$

where $[\mathbf{y}_{\lambda}\mathbf{z}_{\mu}]$ denotes the coefficient of the monomial $\mathbf{y}_{\lambda}\mathbf{z}_{\mu}$. Since Thrall's problem and understanding plethysm coefficients are longstanding open problems in algebraic combinatorics, this is as far as we are able to simplify our answer for now.

4.3. EXAMPLE OF MAIN THEOREM. As an example of Theorem 5.39, we analyze the case n = 4 in the table below. The box in row ν and column μ is filled with $[(\mathbb{C}\mathcal{F}_4 E_{\mu})^{\nu} : M_{\lambda}]$ copies of λ , and the numbers in parentheses indicate multiplicities.



To illustrate how to fill this table, we work out the Σ_4 - composition factors of $(\mathbb{C}\mathcal{F}_4 E_{211})^{\nu}$ as ν varies over partitions of 4, i.e. the shaded boxes. Although this is the most technical column of the table, it is also the most rich. We invite the reader try the other columns using the same process.

Each term in the expansion of the generating function

$$\prod_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{\text{partition} \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_w]$$

is formed by choosing one (potentially empty) partition ρ for each Lyndon word factor w and multiplying the corresponding terms $\mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w|\rho|} L_{\rho}[h_w]$. To obtain terms with \mathbf{z} -weight $\mathbf{z}_{211} = z_2 z_1^2$, the only Lyndon word factors w from which one can choose a nonempty partition are w = 1, w = 2, w = 12, and w = 112. Writing these relevant factors first, our generating function is:

$$\underbrace{\left(\sum_{\rho} \mathbf{y}_{\rho} \mathbf{z}_{1}^{|\rho|} L_{\rho}[h_{1}]\right) \left(\sum_{\rho} \mathbf{y}_{\rho \cdot 2} \mathbf{z}_{2}^{|\rho|} L_{\rho}[h_{2}]\right) \left(\sum_{\rho} \mathbf{y}_{\rho \cdot 3} \mathbf{z}_{12}^{|\rho|} L_{\rho}[h_{12}]\right) \left(\sum_{\rho} \mathbf{y}_{\rho \cdot 4} \mathbf{z}_{112}^{|\rho|} L_{\rho}[h_{112}]\right)}_{w=1} \dots$$

Labelling each term by the factors w for which a nonempty ρ was chosen, the coefficient of \mathbf{z}_{211} in the generating function is

$$\underbrace{\underbrace{\mathbf{y}_{2}L_{2}[h_{1}]}_{\substack{w=1\\\rho=2}} \cdot \underbrace{\mathbf{y}_{2}L_{1}[h_{2}]}_{\substack{w=2\\\rho=1}} + \underbrace{\mathbf{y}_{11}L_{11}[h_{1}]}_{\substack{w=1\\\rho=1}} \cdot \underbrace{\mathbf{y}_{2}L_{1}[h_{2}]}_{\substack{w=2\\\rho=1}} + \underbrace{\mathbf{y}_{1}L_{1}[h_{1}]}_{\substack{w=1\\\rho=1}} \cdot \underbrace{\mathbf{y}_{3}L_{1}[h_{1}]}_{\substack{w=1\\\rho=1}} + \underbrace{\mathbf{y}_{4}L_{1}[h_{12}]}_{\substack{w=1\\\rho=1}} + \underbrace{\mathbf{y}$$

 $Expanding^{(8)}$ into Schur functions, we obtain:

(12)
$$\mathbf{y}_{22}(s_{211}+s_{31}) + \mathbf{y}_{211}(s_4+s_{22}+s_{31}) + \mathbf{y}_{31}(s_4+2s_{31}+s_{22}+s_{211}) \\ + \mathbf{y}_4(s_4+2s_{31}+s_{22}+s_{211}).$$

This process reveals how to fill each box of the shaded column. For instance, the composition multiplicity of M_4 in $(\mathbb{CF}_4E_{211})^{31}$ is $6 = 3 \cdot 2$ because $f^{(3,1)} = 3$ and the coefficient of $\mathbf{y}_4 s_{31}$ in Eq. (12) is 2, as indicated by the blue coloring above.

Algebraic Combinatorics, Vol. 8 #2 (2025)

 $^{^{(8)}}$ One can do this with SageMath, since plethysm and the higher Lie characters (under the name gessel_reutenauer symmetric functions) are implemented.

4.4. RECOVERING RESULTS OF GARSIA-REUTENAUER AND UYEMURA-REYES. As further examples, we explain how Theorem 5.39 specializes to recover results of Garsia-Reutenauer in [25] and Uyemura-Reyes in [45].

4.4.1. The bottom row ($\nu = n$): Garsia-Reutenauer's Cartan invariants of Σ_n . In [25, Theorem 5.4], Garsia and Reutenauer computed the Cartan invariants⁽⁹⁾ of the descent algebra. Let type(α) for a composition α be the partition obtained by reordering $|w_1|, |w_2|, \ldots, |w_k|$ where $w_1w_2 \ldots w_k$ is the unique factorization of α into weakly decreasing (lexicographically) Lyndon words (see Section 4.1).

THEOREM 4.1 (Garsia-Reutenauer). The composition multiplicity of the Σ_n -simple module M_{λ} in the projective indecomposable module P_{μ} is

$$[P_{\mu}: M_{\lambda}] = \#\{\alpha \sim \mu : \operatorname{type}(\alpha) = \lambda\}.$$

EXAMPLE 4.2. The compositions rearranging to (2, 1, 1) are (2, 1, 1), (1, 2, 1), and (1, 1, 2). By taking the Lyndon types of these compositions, as the table below illustrates, we obtain that the composition factors of P_{211} are one copy each of M_{211} , M_{31} , and M_4 . Compare this to the box in row (4) and column (2, 1, 1) of the table in Section 4.3.

α	Lyndon Factorization	$type(\alpha)$
(2, 1, 1)	(2)(1)(1)	(2, 1, 1)
(1, 2, 1)	(1,2)(1)	(3,1)
(1, 1, 2)	(1, 1, 2)	(4)

As right descent algebra modules, $P_{\mu} = \Phi^{-1}(E_{\mu})\Sigma_n \cong (\mathbb{C}\mathcal{F}_n)^n E_{\mu} = (\mathbb{C}\mathcal{F}_n E_{\mu})^n$ by Eq. (3). So, by Eq. (11),

$$[P_{\mu}: M_{\lambda}] = \left\langle s_n, [\mathbf{y}_{\lambda} \mathbf{z}_{\mu}] \prod_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{\text{partition} \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_w] \right\rangle.$$

We show Theorem 5.39 recovers Garsia–Reutenauer's result by computing the composition multiplicity $[P_{\mu} : M_{\lambda}]$ using the above equation. To do so, we will need Lemma 4.4. Our proof of Lemma 4.4 uses the following fact (appearing as Lemma 2.25 in [3]) which will also be useful for a proof later on.

LEMMA 4.3 ([3, Lemma 2.25]). Let H be a subgroup of G. Let W be an H-representation and V be an S_n -representation. Then, as $S_n[G]$ representations,

$$V[W] \Big|_{S_n[H]}^{S_n[G]} \cong V \left[W \Big|_{H}^G \right].$$

LEMMA 4.4. Let ν, μ be partitions of j, k respectively. Set $n = j \cdot k$. Then,

$$\langle s_n, L_{\nu}[h_{\mu}] \rangle = \begin{cases} 1 & \text{if } \nu = 1^j, \\ 0 & \text{otherwise.} \end{cases}$$

Proof. We first show that $\sum_{\nu \vdash j} \langle s_n, L_{\nu}[h_{\mu}] \rangle = 1$. By linearity of the Hall inner product, Lemma 2.10(2), and Eq. (9),

$$\sum_{\nu \vdash j} \langle s_n, L_{\nu}[h_{\mu}] \rangle = \left\langle s_n, \sum_{\nu \vdash j} L_{\nu}[h_{\mu}] \right\rangle = \left\langle s_n, \left(\sum_{\nu \vdash j} L_{\nu} \right) [h_{\mu}] \right\rangle = \langle s_n, h_{1^j}[h_{\mu}] \rangle.$$

⁽⁹⁾They actually proved a stronger result by finding a basis for each space $\Phi^{-1}(E_{\mu})\Sigma_n\Phi^{-1}(E_{\lambda})$.

By Lemma 2.10(3) and the fact that $h_1[f] = f$, we can simplify the above to

$$\left\langle s_n, (h_1[h_\mu])^j \right\rangle = \left\langle s_n, (h_\mu)^j \right\rangle = \left\langle s_n, h_{\mu j} \right\rangle.$$

Eq. (7) then implies that

$$\sum_{\nu\vdash j} \left\langle s_n, L_{\nu}[h_{\mu}] \right\rangle = \left\langle s_n, h_{\mu^j} \right\rangle = K_{n,\mu^j} = 1.$$

Since $L_{\nu}[h_{\mu}]$ is the Frobenius characteristic of an S_n -representation, namely $\mathcal{L}_{\nu}\left[\mathbbm{1}\uparrow_{S_{\mu}}^{S_k}\right]\uparrow_{S_j[S_k]}^{S_n}$, we know that each $L_{\nu}[h_{\mu}]$ is a nonnegative integer linear combination of Schur functions. Thus, there is exactly one partition $\nu \vdash j$ for which $\langle s_n, L_{\nu}[h_{\mu}] \rangle$ is 1, and $\langle s_n, L_{\nu}[h_{\nu}] \rangle$ must be 0 otherwise. So, to prove the lemma, it suffices to prove that $\langle s_n, L_{(1^j)}[h_\mu] \rangle = 1$.

Since $L_{1^j} = h_j[L_1] = h_j$, the inner product $\langle s_n, L_{1^j}[h_\mu] \rangle = \langle s_n, h_j[h_\mu] \rangle$. By Lemma 4.3 and transitivity of induction, $h_j[h_\mu]$ is the Frobenius image of

$$\mathbb{1}\left[\mathbb{1}\uparrow_{S_{\mu}}^{S_{k}}\right]\uparrow_{S_{j}[S_{k}]}^{S_{n}}\cong\mathbb{1}\left[\mathbb{1}\right]\uparrow_{S_{j}[S_{\mu}]}^{S_{j}[S_{\mu}]}\uparrow_{S_{j}[S_{k}]}^{S_{n}}\cong\mathbb{1}\left[\mathbb{1}\right]\uparrow_{S_{j}[S_{\mu}]}^{S_{n}}\cong\mathbb{1}\uparrow_{S_{j}[S_{\mu}]}^{S_{n}}$$

which is the coset representation on $S_n/S_j[S_\mu]$. As a transitive permutation representation, it contains the the trivial representation with multiplicity one, completing the lemma.

PROPOSITION 4.5. Theorem 5.39 recovers Garsia-Reutenauer's Cartan invariants of Σ_n . Namely,

$$\left\langle s_n, [\mathbf{y}_{\lambda} \mathbf{z}_{\mu}] \prod_{\substack{Lyndon\\w}} \sum_{\substack{partition\\\rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_w] \right\rangle = \#\{\alpha \sim \mu : \operatorname{type}(\alpha) = \lambda\}.$$

Proof. Observe that

$$\left[\mathbf{y}_{\lambda}\mathbf{z}_{\mu}\right]\prod_{\substack{\text{Lyndon}\\w}}\sum_{\substack{\text{partition}\\\rho}}\mathbf{y}_{\rho\cdot|w|}\mathbf{z}_{w|\rho|}L_{\rho}[h_{w}] = \sum_{(\rho^{w})}\left[\mathbf{y}_{\lambda}\mathbf{z}_{\mu}\right]\left(\prod_{w}\mathbf{y}_{\rho^{w}\cdot|w|}\mathbf{z}_{w|\rho^{w}|}L_{\rho^{w}}[h_{w}]\right).$$

where the sum on the right side is over vectors (ρ^w) of partitions (one partition ρ^w for each Lyndon word w) with only finitely many of the partitions nonempty. Extracting the coefficient of $\mathbf{y}_{\lambda} \mathbf{z}_{\mu}$ from each term, we simplify the above to

$$\sum_{(\rho^w)\in S(\lambda,\mu)}\prod_w L_{\rho^w}[h_w],$$

where $S(\lambda, \mu)$ is the set of partition vectors (ρ^w) for which

- the concatenation of the partitions ρ^w · |w| rearranges to λ, and
 the concatenation of the partitions w^{|ρ^w|} rearranges to μ.

By decomposing each f_i into Schur functions and applying the Littlewood–Richardson rule, one can show that if f_1, f_2, \ldots, f_k are homogeneous symmetric functions of degree m_1, m_2, \ldots, m_k and $n = \sum_{i=1}^k m_i$, then

$$\left\langle s_n, \prod_{i=1}^k f_i \right\rangle = \prod_{i=1}^k \left\langle s_{m_i}, f_i \right\rangle.$$

Applying this fact, we have

Algebraic Combinatorics, Vol. 8 #2 (2025)

$$\sum_{w)\in S(\lambda,\mu)} \left\langle s_n, \prod_w L_{\rho^w}[h_w] \right\rangle = \sum_{(\rho^w)\in S(\lambda,\mu)} \prod_w \left\langle s_{(|\rho^w|\cdot|w|)}, L_{\rho^w}[h_w] \right\rangle.$$

By Lemma 4.4, for a fixed $(\rho^w) \in S(\lambda, \mu)$,

 $(\rho$

$$\prod_{w} \left\langle s_{(|\rho^{w}| \cdot |w|)}, L_{\rho^{w}}[h_{w}] \right\rangle = \begin{cases} 1 & \text{if } \rho^{w} = 1^{|\rho^{w}|} \text{ for each } w, \\ 0 & \text{otherwise.} \end{cases}$$

Hence, $\sum_{(\rho^w)\in S(\lambda,\mu)} \prod_w \left\langle s_{(|\rho^w|\cdot|w|)}, L_{\rho^w}[h_w] \right\rangle \text{ counts the number of ways to pick a}$

vector $(\operatorname{mult}(w))_w$ of nonnegative integers (one integer $\operatorname{mult}(w)$ for each Lyndon word w with only finitely many $\operatorname{mult}(w) \neq 0$) so that the vector $(1^{\operatorname{mult}(w)}) \in S(\lambda, \mu)$.

The set of such vectors is in bijection with the set $\{\alpha \sim \mu : \operatorname{type}(\alpha) = \lambda\}$. Specifically, consider some vector $(\operatorname{mult}(w))_w$ of nonnegative integers satisfying the described conditions. The composition obtained by concatenating $\operatorname{mult}(w)$ copies of each Lyndon word w for which $\operatorname{mult}(w) \neq 0$ (in lexicographically weakly decreasing order) is necessarily a composition in the set $\{\alpha \sim \mu : \operatorname{type}(\alpha) = \lambda\}$. For example, if $\operatorname{mult}(112) = 2$, $\operatorname{mult}(14) = 3$, and $\operatorname{mult}(w) = 0$ for all other w, then the vector $(1^{\operatorname{mult}(w)})$ is in $S(5^{3}4^{2}, 4^{3}2^{2}1^{7})$ and its corresponding composition 141414112112 rearranges to $4^{3}2^{2}1^{7}$ and has Lyndon type $5^{3}4^{2}$. Since Lyndon factorization is unique and always exists, this map defines a bijection, completing the proof. \Box

4.4.2. The rightmost column ($\mu = 1^n$): Uyemura-Reyes's shuffling representations. It is simple to check that the coefficient of \mathbf{z}_{1^n} in Eq. (10) is

$$\sum_{\lambda \vdash n} \mathbf{y}_{\lambda} L_{\lambda}[h_1] = \sum_{\lambda \vdash n} \mathbf{y}_{\lambda} L_{\lambda}.$$

Hence, Theorem 5.39 reveals that

(13)
$$[(\mathbb{C}\mathcal{F}_n E_{1^n})^{\nu} : M_{\lambda}] = f^{\nu} \cdot \langle s_{\nu}, L_{\lambda} \rangle.$$

This turns out to recover a result of Uyemura–Reyes which was originally conjectured by R. Stanley. In particular, since Σ_n is an elementary algebra, the (potentially non-distinct) eigenvalues of an element $x \in \Sigma_n$ acting on the *right* of the S_n -isotypic subspace $(\mathbb{C}S_n)^{\nu}$ are indexed by partitions $\lambda \vdash n$. The eigenvalue associated to λ is $\chi_{M_{\lambda}}(x)$, where $\chi_{M_{\lambda}}$ is the character associated to the simple Σ_n -representation M_{λ} and has multiplicity $[(\mathbb{C}S_n)^{\nu} : M_{\lambda}]$; see [2, Theorem D.38]. In [45, Theorem 4.1], Uyemura–Reyes studied the S_n –representation-structure of the eigenspace associated to λ for an element $x \in \Sigma_n$ acting semisimply on $\mathbb{C}S_n$. His work gives that the Frobenius image of the λ -eigenspace is L_{λ} . Hence,

(14)
$$[(\mathbb{C}S_n)^{\nu}: M_{\lambda}] = f^{\nu} \cdot \langle s_{\nu}, L_{\lambda} \rangle.$$

Eq. (13) and Eq. (14) are actually equivalent. Specifically, it turns out out that $\mathbb{C}\mathcal{F}_n E_{1^n}$ is equal to the \mathbb{C} -span of the **chambers** (top-dimensional faces) of \mathcal{B}_n . (This follows from the fact that E_{1^n} can be written as the sum $\sum_{X \in 1^n} E_X$ and each E_X is a sum of chambers by Lemma 2.5(ii).) The symmetric group acts simply transitively on the chambers which provides an S_n -isomorphism between the chamber space and the group algebra. Under this S_n -isomorphism, acting by x on the right of $\mathbb{C}S_n$ is equivalent to acting on the left of the chamber space by $\Phi^{-1}(x)$ (see, for instance, [20, Theorem 8]).

In [43, Theorem 4.2], Reiner–Saliola–Welker generalize the result of Uyemura– Reyes to all Coxeter groups in terms of the (twisted) cohomology of lower intervals in the lattice of intersections. Their result recovers the case $\mu = 1^n$ in Corollary 5.9 and more generally $[K] = [\emptyset]$ in Proposition 6.2.

4.5. MORE EXPLICIT ANSWER FOR THE SIGN ISOTYPIC SUBSPACE. Although Theorem 5.39 is as complete of an answer as we can provide for most isotypic subspaces, we are able to provide a more explicit answer for the sign isotypic subspace.

PROPOSITION 4.6. As Σ_n -modules, the sign isotypic subspace $(\mathbb{C}\mathcal{F}_n)^{1^n}$ of the face algebra is isomorphic to M_{λ} where $\lambda = (2^{\frac{n}{2}})$ if n is even and $\lambda = (2^{\frac{n-1}{2}}, 1)$ if n is odd.

Proof. By Proposition 3.3 the dimension of $(\mathbb{C}\mathcal{F}_n E_\mu)^{1^n}$ is $\#\{\alpha \vDash n : \alpha \sim \mu\} \cdot K_{1^n,\mu}$, which is one if $\mu = 1^n$ and 0 otherwise. Thus, $(\mathbb{C}\mathcal{F}_n)^{1^n} = (\mathbb{C}\mathcal{F}_n E_{1^n})^{1^n}$ and is one dimensional. To determine the Σ_n -isomorphism type of $(\mathbb{C}\mathcal{F}_n)^{1^n} = (\mathbb{C}\mathcal{F}_n E_{1^n})^{1^n}$, it suffices to find the unique $\lambda \vdash n$ for which $\left[(\mathbb{C}\mathcal{F}_n E_{1^n})^{1^n} : M_\lambda \right]$ is nonzero. By Eq. (13),

$$\left[\left(\mathbb{C}\mathcal{F}_{n}E_{1^{n}}\right)^{1^{n}}:M_{\lambda}\right]=\left\langle s_{1^{n}},L_{\lambda}\right\rangle.$$

A special case of a beautiful result of Gessel–Reutenauer (see [26, Theorem 2.1]) explains that $\langle L_{\lambda}, s_{1^n} \rangle$ counts permutations in S_n with cycle type λ and descent set $\{1, 2, \ldots, n-1\}$. Hence, the scalar product is zero except for when λ is the cycle type of the longest word of S_n , which is well known to be $(2^{\frac{n}{2}})$ for n even and $(2^{\frac{n-1}{2}}, 1)$ for n odd.

5. Proof of the main theorem

5.1. OUTLINE OF PROOF. The proof of Theorem 5.39 is quite long. There are also portions which may be of interest in their own right, involving the homology of intervals in the set partition lattice and the interpretation of L_n in terms of necklaces. In this section, we outline the main ideas in an attempt to (1) make our involved proof more digestible and (2) guide the curious reader towards these ingredients of potential interest. All gaps in this proof outline will be filled in Sections 5.2, 5.3, 5.4, and 5.5.

Step 1. Reduction to homology of intervals in the set partition lattice. Let $\operatorname{Stab}_{S_n}(X)$ denote the S_n -stabilizer subgroup of the set partition X. We shall prove the following proposition.

PROPOSITION 5.2. If μ does not refine λ , then $E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu} = 0$. Otherwise, as S_n -representations,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X \leqslant Y]} E_{Y}\mathbb{C}\mathcal{F}_{n}E_{X} \Big|_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}},$$

where the direct sum is over S_n -orbits of pairs $X \leq Y$ in Π_n with $X \in \mu, Y \in \lambda$.

Proposition 5.2 reduces our proof to understanding the $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ representations $E_Y \mathbb{CF}_n E_X$. A twisting character appears when studying these representations. The set partition lattice Π_n is $(S_n$ -equivariantly) isomorphic to the lattice
of intersections of the hyperplanes of \mathcal{B}_n . Define det(Y) be the ± 1 -valued $\operatorname{Stab}_{S_n}(Y)$ character sending $w \in \operatorname{Stab}_{S_n}(Y)$ to its determinant when restricted to the subspace
associated to Y. For more details about this determinant character, see Section 5.2.2.

Saliola explicitly proves the non-equivariant version of the following lemma in [49, §10.2], and the determinant twists making it equivariant appear implicitly in his work in [48, Theorem 6.2]. Aguiar and Mahajan also state it in a more general context in [2,

Proposition 14.44]. We will include a proof for completeness. Let the **length** $\ell(X)$ of a set partition mean the number of blocks.

LEMMA 5.8. Assume $X, Y \in \Pi_n$ with X refining Y and $\ell(X) - \ell(Y) = k$. Then, as representations of the subgroup $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$,

$$E_Y \mathbb{C} \mathcal{F}_n E_X \cong \tilde{H}^{k-2}(X,Y) \otimes \det(Y) \otimes \det(X),$$

where $\tilde{H}^{k-2}(X,Y)$ is the reduced cohomology of the order complex of the open interval (X,Y) in Π_n .

Note that we shall explain our poset topology conventions in Section 5.2.4. Combining Proposition 5.2, Lemma 5.8, and properties of dual representations, we prove the following reformulation. Analogously to the length of a set partition, let the **length** $\ell(\mu)$ of an integer partition μ be the number of its parts.

COROLLARY 5.9. If μ does not refine λ , then $E_{\lambda}\mathbb{CF}_{n}E_{\mu} = 0$. Otherwise, let $\ell(\mu) - \ell(\lambda) = k$. Then, as S_{n} -representations,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X \leqslant Y]} \left(\tilde{H}_{k-2}\left(X,Y\right) \otimes \det(Y) \otimes \det(X) \right) \Big|_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}}$$

where the direct sum is over S_n -orbits of pairs $X \leq Y$ in Π_n with $X \in \mu, Y \in \lambda$ and \tilde{H}_{k-2} means the reduced homology.

The proofs of Proposition 5.2, Lemma 5.8, and Corollary 5.9 appear in Section 5.2.

Step 2: Understand upper intervals. The next key step in proving Theorem 5.39 is understanding ch $(E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu})$ in the special case $\lambda = n$. We do so by proving the following proposition.

Proposition 5.1.

$$\sum_{\mu \neq \varnothing} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(E_{|\mu|} \mathbb{C} \mathcal{F}_{|\mu|} E_{\mu} \right) = \sum_{\substack{Lyndon \\ w}} \sum_{m \ge 1} \mathbf{z}_{w^m} \cdot L_m[h_w].$$

Indeed, if $\{X_{\mu} : \mu \vdash n\}$ are set partitions where X_{μ} has block sizes μ , then by Corollary 5.9, the left side of Proposition 5.1 can be rewritten as

(15)
$$\sum_{\mu \neq \emptyset} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(\tilde{H}_{\ell(\mu)-1} \left(X_{\mu}, \hat{1} \right) \otimes \det \left(X_{\mu} \right) \uparrow_{\operatorname{Stab}_{S_{|\mu|}}(X_{\mu})}^{S_{|\mu|}} \right).$$

Sundaram studied the homology of the partition lattice in great depth in [59]. In [59, proof of Thm 1.4], she studies the $\operatorname{Stab}_{S_{|\mu|}}(X_{\mu})$ -representations $\tilde{H}_{\ell(\mu)-1}(X_{\mu}, \hat{1})$. By adjusting her work with the appropriate determinant twists, we can prove the next lemma. (Note that plethysm of formal power series is explained in Section 5.3.2.)

Lemma 5.18.

$$\sum_{\mu\neq\varnothing} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(E_{|\mu|} \mathbb{C} \mathcal{F}_{|\mu|} E_{\mu} \right) = \sum_{r\geqslant 1} L_r [z_1 h_1 + z_2 h_2 + \ldots].$$

The symmetric functions L_r can be written as a generating function counting combinatorial objects called primitive necklaces. Using this interpretation, we construct a technical necklace bijection to prove the following lemma.

Lemma 5.25.

$$\sum_{r \ge 1} L_r[z_1h_1 + z_2h_2 + \ldots] = \sum_{\substack{Lyndon \\ w}} \sum_{m \ge 1} \mathbf{z}_{w^m} L_m[h_w].$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

To complete the proof of Proposition 5.1 it suffices to provide proofs of Lemma 5.18 and Lemma 5.25, which we do in Section 5.3.

Step 3: Use upper intervals to understand general intervals: By identifying the intersections of the subgroups $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ (Lemma 5.32), we recast their action on intervals of the partition lattice as the action of (wreath) products of smaller subgroups on products of smaller partition lattices (with appropriate determinant twists). Using Sundaram's work in [58], we can express the twisted homology representations $\tilde{H}_{k-2}(X,Y) \otimes \det(X) \otimes \det(Y)$ in terms of the upper intervals from Step 2 (Lemma 5.35). All of these arguments appear in Section 5.4.

Step 4: Proof of main theorem: We use Step 3 to obtain the Frobenius characteristic of $E_{\lambda} \mathbb{CF}_n E_{\mu}$ in Proposition 5.37. With this result, we complete the proof of our main theorem (Theorem 5.39) using a generating function argument. These two arguments appear in Section 5.5.

5.2. REDUCTION TO HOMOLOGY OF INTERVALS IN THE LATTICE OF SET PARTITIONS. In this subsection we fill in the gaps to Step 1 of the proof outline in Section 5.1. We prove Proposition 5.2 in Section 5.2.1. We then give background on the determinant character in Section 5.2.2, on a special map defined by Saliola in Section 5.2.3, and on poset topology in Section 5.2.4, so that we can prove Lemma 5.8 and Corollary 5.9 in Section 5.2.5.

5.2.1. Proof of Proposition 5.2.

PROPOSITION 5.2. If μ does not refine λ , then $E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu} = 0$. Otherwise, as S_n -representations,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X \leqslant Y]} E_{Y}\mathbb{C}\mathcal{F}_{n}E_{X} \Big\uparrow^{S_{n}}_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}$$

where the direct sum is over S_n -orbits of pairs $X \leq Y$ in Π_n with $X \in \mu, Y \in \lambda$.

Proof. Since $E_{\lambda} = \sum_{Y \in \lambda} E_Y$, $E_{\mu} = \sum_{X \in \mu} E_X$, and the idempotents $\{E_X : X \in \Pi_n\}$ are orthogonal, there is a vector space decomposition

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} = \bigoplus_{\substack{Y \in \lambda, \\ X \in \mu}} E_{Y}\mathbb{C}\mathcal{F}_{n}E_{X}.$$

By Lemma 2.5(iii), the spaces $E_Y \mathbb{C} \mathcal{F}_n E_X$ are zero unless $X \leq Y$. Hence,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} = \bigoplus_{X \leqslant Y} E_{Y}\mathbb{C}\mathcal{F}_{n}E_{X},$$

where the direct sum is over pairs $X \leq Y$ in Π_n with $X \in \mu$, $Y \in \lambda$. Observe that $w(E_Y \mathbb{C} \mathcal{F}_n E_X) = E_{w(Y)} \mathbb{C} \mathcal{F}_n E_{w(X)}$ for $w \in S_n$, so the direct summands above are permuted by S_n . The S_n -stabilizer of the summand $E_Y \mathbb{C} \mathcal{F}_n E_X$ is $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$. The proposition then follows from standard theory of induced representations (see, for instance, [66, Proposition 4.3.2]).

5.2.2. The determinant character. Each set partition $X = \{X_1, X_2, \ldots, X_k\}$ of [n] into k parts is associated to a k-dimensional subspace of \mathbb{R}^n . In particular, letting e_1, e_2, \ldots, e_n be the standard basis vectors for \mathbb{R}^n , X is associated to the space spanned by the vectors v_1, v_2, \ldots, v_k , where

$$v_i \coloneqq \sum_{j \in X_i} e_j$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

An **orientation** of a subspace V of \mathbb{R}^n is some choice of an ordered basis \mathcal{V} for V. Given this choice, an ordered basis \mathcal{V}' of V is said to be **positive** if the change of basis matrix between \mathcal{V} and \mathcal{V}' has positive determinant and **negative** otherwise.

A permutation $w \in S_n$ maps an ordered basis of the subspace associated to set partition X to an ordered basis of the subspace associated to wX. If one fixes an orientation on each subspace of \mathbb{R}^n associated to a set partition of Π_n , define $\sigma_X(w)$ to be +1 if w maps a positively oriented basis of the space associated to X to a positively oriented basis of the space associated to wX and -1 otherwise.

Crucially, note that when $w \in \operatorname{Stab}_{S_n}(X)$, then $\sigma_X(w)$ does not depend on the choice of orientation. In this situation, $\sigma_X(w) = \det(X)(w)$. (Recall that the **determinant character** $\det(X)$ is the ± 1 -valued $\operatorname{Stab}_{S_n}(X)$ character sending w to the determinant of its action on the subspace associated to X). We provide an example of $\det(X)$ below.

EXAMPLE 5.3. Let $X = \{15, 26, 37, 48\}$. One choice of orientation for the subspace of \mathbb{R}^8 associated to X is (v_1, v_2, v_3, v_4) with $v_i = e_i + e_{i+4}$. Using cycle notation, the permutation w = (1, 5)(3, 8)(4, 7)(2, 6) lives in $\operatorname{Stab}_{S_n}(X)$ and (expressed with respect to the chosen orientation) has matrix $\begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \end{bmatrix}$. Since this matrix has determinant -1, $\det(X)(w) = -1$, even though $\operatorname{sgn}(w) = \det(\mathbb{R}^8)(w) = +1$.

5.2.3. Saliola's map. We assume some basic familiarity with quivers and path algebras in this subsection; see [48, §3] for more details.

In [48, 49], Saliola proved the remarkable fact that the **quiver** Q_n of the face algebra $\mathbb{C}\mathcal{F}_n$ is the (directed) Hasse diagram of the set partition lattice Π_n . Namely, the vertices ε_X of Q_n are indexed by set partitions X in Π_n and there is an arrow $\varepsilon_X \to \varepsilon_Y$ if and only if Y covers X in Π_n . The **path algebra** $\mathbb{C}Q_n$ has all paths in Q_n as a \mathbb{C} -basis; such paths correspond to unrefinable chains of Π_n . Given choices of orientations on each subspace associated to a set partition, Saliola defines an action of S_n on $\mathbb{C}Q_n$. In particular, for $w \in S_n$

(16)

$$w\left(\varepsilon_{X_1}\to\varepsilon_{X_2}\to\ldots\to\varepsilon_{X_k}\right)\coloneqq\sigma_{X_k}(w)\left(\varepsilon_{w(X_1)}\to\varepsilon_{w(X_2)}\to\ldots\to\varepsilon_{w(X_k)}\right).$$

By the definition of multiplication in the path algebra, the \mathbb{C} -vector space $\varepsilon_Y \mathbb{C} \mathcal{Q}_n \varepsilon_X$ has as a \mathbb{C} -basis all paths $\varepsilon_X \to \ldots \to \varepsilon_Y$ in \mathcal{Q}_n , which correspond with unrefinable chains from X to Y in Π_n . Hence, for $X \leq Y$ in Π_n , the space $\varepsilon_Y \mathbb{C} \mathcal{Q}_n \varepsilon_X$ is closed under the action of $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$. Importantly, since $\sigma_X(w)\sigma_Y(w) =$ $\det(X)(w) \det(Y)(w)$ for $w \in \operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$, the action of $\operatorname{Stab}_{S_n}(X) \cap$ $\operatorname{Stab}_{S_n}(Y)$ on $\varepsilon_Y \mathbb{C} \mathcal{Q}_n \varepsilon_X$ is *independent* of the choices of orientation.

In [48, Theorem 6.2], Saliola defines an explicit, surjective algebra homomorphism

$$\varphi:\mathbb{C}\mathcal{Q}_n\twoheadrightarrow\mathbb{C}\mathcal{F}_n$$

We will not need φ 's precise definition for our purposes, but we will use two of its properties, listed below (see [48, Theorem 6.2]). Recall from Section 2.4.3 that $\{\widehat{E}_X : X \in \Pi_n\}$ are Saliola's explicit idempotents⁽¹⁰⁾ for $\mathbb{C}\mathcal{F}_n$.

- (i) $\varphi(\varepsilon_X) = \widehat{E}_X$.
- (ii) φ is S_n -equivariant with respect to the action in Eq. (16).

⁽¹⁰⁾Saliola's map can be adjusted to work for the general $\{E_X\}$ fixed at the end of Section 2.4.3. However, to avoid having to explain his proof, we circumvent this issue by sticking with his idempotents and connecting them back to the fixed system $\{E_X\}$ in Lemma 5.7

Since ker $\varphi|_{\varepsilon_Y \mathbb{C}Q_n \varepsilon_X} = \varepsilon_Y$ (ker $\varphi) \varepsilon_X$, there is an isomorphism of $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ - representations:

(17)
$$\varepsilon_Y \mathbb{C} \mathcal{Q}_n \varepsilon_X / \varepsilon_Y (\ker \varphi) \varepsilon_X \longrightarrow \widehat{E}_Y \mathbb{C} \mathcal{F}_n \widehat{E}_X$$
$$p + \varepsilon_Y (\ker \varphi) \varepsilon_X \mapsto \varphi(p).$$

In [48, Theorem 6.2], Saliola proves the kernel of φ is the two-sided ideal of $\mathbb{C}Q_n$ generated by the sum of all the paths of length two – we shall call this special sum s. Write \ll to denote the **covering** relation in Π_n .

LEMMA 5.4. ε_Y (ker φ) ε_X is the \mathbb{C} -span of the elements

$$x_C \coloneqq \sum_{X_i \leqslant Z \leqslant X_{i+1}} \varepsilon_X \to \varepsilon_{X_1} \to \ldots \to \varepsilon_{X_i} \to \varepsilon_Z \to \varepsilon_{X_{i+1}} \to \ldots \to \varepsilon_{X_k} \to \varepsilon_Y$$

as one varies over chains C of the form

$$C = (X \lessdot X_1 \lessdot \ldots \lessdot X_i \lt X_{i+1} \lessdot \ldots \lessdot X_k \lessdot Y)$$

in Π_n where $\ell(X_{i+1}) - \ell(X_i) = 2$.

Proof. Observe that

$$\sum_{X_i < Z < X_{i+1}} \varepsilon_X \to \varepsilon_{X_1} \to \dots \to \varepsilon_{X_i} \to \varepsilon_Z \to \varepsilon_{X_{i+1}} \to \dots \to \varepsilon_{X_k} \to \varepsilon_Y$$
$$= \left(\varepsilon_{X_{i+1}} \to \dots \to \varepsilon_{X_k} \to \varepsilon_Y\right) \cdot s \cdot \left(\varepsilon_X \to \varepsilon_{X_1} \to \dots \to \varepsilon_{X_i}\right).$$

Hence, each x_C is in ε_Y (ker φ) ε_X . Any element of ε_Y (ker φ) ε_X can be written as

$$\left(\sum_{y} c_{y} y\right) \cdot s \cdot \left(\sum_{x} c_{x} x\right) = \sum_{x,y} c_{x} c_{y} (y \cdot s \cdot p),$$

where the sums are over paths x beginning at ε_X and paths y ending at Y. The element $(y \cdot s \cdot p)$ is zero if the end of path x and the start of path y are not connected by a path of length two and otherwise is one of the x_C , completing the proof.

If X = Y, the restriction of ker φ to $\varepsilon_Y \mathbb{C} \mathcal{Q} \varepsilon_X$ is empty, so as $\operatorname{Stab}_{S_n}(X)$ representations, $\widehat{E}_X \mathbb{C} \mathcal{F}_n \widehat{E}_X \cong \varepsilon_X \mathbb{C} \mathcal{Q} \varepsilon_X \cong \operatorname{span}_{\mathbb{C}} \{\varepsilon_X\}$, the trivial representation.

5.2.4. Poset homology and cohomology. We give a (very brief) recollection of poset homology and cohomology here. For more details, see [65] as an excellent source. We will be particularly interested in the topology of *open intervals* within posets. Throughout, assume that $X \leq Y$ in some ambient poset P. We shall define a chain complex

$$\dots \xrightarrow{\partial_3} C_2(X,Y) \xrightarrow{\partial_2} C_1(X,Y) \xrightarrow{\partial_1} C_0(X,Y) \xrightarrow{\partial_0} C_{-1}(X,Y) \xrightarrow{\partial_{-1}} C_{-2}(X,Y) \xrightarrow{\partial_{-2}} \dots,$$

where precise definition of the \mathbb{C} -vector spaces $C_i(X, Y)$ and boundary maps ∂_i depends on whether we are in the more unusual situation where X = Y or the standard situation where X < Y.

Case 1: X = Y. When X = Y, we set $C_{-2}(X, X) = \mathbb{C}$ and $C_i(X, X) = 0$ for all other *i*. The map ∂_{-2} is defined to be the zero map $\partial_{-2} : C_{-2}(X, Y) = \mathbb{C} \to C_{-3}(X,Y) = 0$.

Case 2: X < Y. An *i*-chain in (X, Y) is a sequence of elements $(X_1 < X_2 < ... < X_{i+1})$, with each X_j in the open interval (X, Y). We define the spaces $C_i(X, Y)$ to be the \mathbb{C} -span of *i*-chains. Note that we allow -1-chains so that $C_{-1}(X, Y) = \mathbb{C}$, the

one-dimensional \mathbb{C} -vector space spanned by the empty chain (). The boundary map $\partial_i : C_i(X,Y) \longrightarrow C_{i-1}(X,Y)$ is the \mathbb{C} -linear map sending

$$(X_1 < X_2 < \ldots < X_{i+1}) \mapsto \sum_{j=1}^{i+1} (-1)^j (X_1 < X_2 < \ldots < X_{j-1} < X_{j+1} < \ldots < X_{i+1}).$$

In both cases, (when $X \leq Y$), the *i*-th reduced homology group of (X, Y), written $\tilde{H}_i(X, Y)$, is defined as

$$\tilde{H}_i(X,Y) := \ker(\partial_i) / \operatorname{im}(\partial_{i+1}).$$

We now define the dual notion of poset *cohomology*. Using the spaces $C_i(X, Y)$ given above, we define a (co-)chain complex

$$\dots \xleftarrow{\partial_2^*} C_2(X,Y) \xleftarrow{\partial_1^*} C_1(X,Y) \xleftarrow{\partial_0^*} C_0(X,Y) \xleftarrow{\partial_{-1}^*} C_{-1}(X,Y) \xleftarrow{\partial_{-2}^*} C_{-2}(X,Y) \xleftarrow{\partial_{-3}^*} \dots$$

where the definition of the boundary map ∂_i^* requires us to once again split into cases. **Case 1:** X = Y. If X = Y, the map ∂_{-2}^* is the zero map from $\mathbb{C} = C_{-2}(X, Y) \to C_{-1}(X, Y) = 0$.

Case 2: X < Y. For X < Y, the coboundary map $\partial_i^* : C_i(X,Y) \longrightarrow C_{i+1}(X,Y)$ is the \mathbb{C} -linear map sending

$$(X_1 < X_2 < \dots < X_{i+1}) \mapsto \sum_{j=1}^{i+2} \sum_{X_{j-1} < X' < X_j} (-1)^j (X_1 < \dots < X_{j-1} < X' < X_j < \dots < X_{i+1}).$$

In either case, we define the *i*-th reduced cohomology group of (X, Y), written $\tilde{H}^i(X, Y)$ to be

$$\tilde{H}^{i}(X,Y) := \ker(\partial_{i}^{*})/\operatorname{im}(\partial_{i-1}^{*}).$$

Note that our homology and cohomology definitions agree with the ordinary reduced homology or cohomology of the order complex of the open interval (X, Y) in P. If a poset P has an absolute minimum element $\hat{0}$ and an absolute maximum element $\hat{1}$, we use the convention of writing $\tilde{H}_i(P)$ and $\tilde{H}^i(P)$ to denote $\tilde{H}_i(\hat{0}, \hat{1})$ and $\tilde{H}^i(\hat{0}, \hat{1})$, respectively.

REMARK 5.5. Assume P is a graded poset and X < Y in P. Set r-2 as the length of the maximal chains of (X, Y). Then, there is precisely one position in which any (r-3)-chain in $C_{r-3}(X, Y)$ can be refined. Thus, ∂_{r-3}^* sends any (r-3)-chain $X_1 < X_2 < \ldots < X_i < X_{i+1} < \ldots < X_{r-2}$ in (X, Y) to

$$(-1)^i \sum_{X_i \ll X' \ll X_{i+1}} X_1 \ll \ldots \ll X_i \ll X' \ll X_{i+1} \ll \ldots \ll X_{r-2}.$$

Hence, the \mathbb{C} -vector space im (∂_{r-3}^*) is the \mathbb{C} -span of elements of the above form as one varies over all (r-3)-chains of (X, Y).

REMARK 5.6. Let P be a finite poset. If a finite group G acts on the elements of P by poset automorphisms, then $\operatorname{Stab}_G(X) \cap \operatorname{Stab}_G(Y)$ acts on each chain group $C_i(X,Y)$ for any $X \leq Y$ in P. If X < Y, we assume here that $\operatorname{Stab}_G(X) \cap \operatorname{Stab}_G(Y)$ acts trivially on the empty chain spanning $C_{-1}(X,Y)$. Similarly, if X = Y, we assume that $\operatorname{Stab}_G(X)$ acts trivially on $C_{-2}(X,X) = \mathbb{C}$. It can be checked that Gcommutes with the maps ∂_i and ∂_i^* . Therefore, $\tilde{H}_i(X,Y)$ and $\tilde{H}^i(X,Y)$ are both $\operatorname{Stab}_G(X) \cap \operatorname{Stab}_G(Y)$ -representations. As Wachs explains in [65, p31], they are *dual* representations (as we are working over a field, namely \mathbb{C}).

Algebraic Combinatorics, Vol. 8 #2 (2025)

5.2.5. Proofs of Lemma 5.8 and Corollary 5.9. To prove Lemma 5.8, we first need a lemma translating between the representations given by our fixed cfpoi of \mathbb{CF}_n and Saliola's.

LEMMA 5.7. As $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ -representations,

$$\widehat{E}_Y \mathbb{C} \mathcal{F}_n \widehat{E}_X \cong E_Y \mathbb{C} \mathcal{F}_n E_X.$$

Proof. Recall from Lemma 2.7 that there exists an invertible element $u \in (\mathbb{C}\mathcal{F}_n)^{S_n}$ for which $u\widehat{E}_X u^{-1} = E_X$ and $u\widehat{E}_Y u^{-1} = E_Y$. The S_n -equivariant conjugation automorphism on $\mathbb{C}\mathcal{F}_n$ mapping $m \mapsto umu^{-1}$ restricts to a $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ -equivariant isomorphism $\widehat{E}_Y \mathbb{C}\mathcal{F}_n \widehat{E}_X \to E_Y \mathbb{C}\mathcal{F}_n E_X$.

We are now ready to prove Lemma 5.8.

LEMMA 5.8. Let $X \leq Y \in \Pi_n$ with $\ell(X) - \ell(Y) = k$. As representations of $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$,

$$E_Y \mathbb{C} \mathcal{F}_n E_X \cong \tilde{H}^{k-2}(X,Y) \otimes \det(Y) \otimes \det(X).$$

Proof. By Lemma 5.7, it suffices to prove that

$$\widehat{E}_Y \mathbb{C} \mathcal{F}_n \widehat{E}_X \cong \widetilde{H}^{k-2}(X, Y) \otimes \det(Y) \otimes \det(X).$$

First, assume that X = Y so that k = 0. As discussed at the end of Section 5.2.3, $\widehat{E}_X \mathbb{C} \mathcal{F}_n \widehat{E}_X$ carries the structure of the trivial representation of $\operatorname{Stab}_{S_n}(X)$. On the other hand, $\widehat{H}^{-2}(X,X)$ carries the trivial representation of $\operatorname{Stab}_{S_n}(X)$, as discussed in Remark 5.6. Since $\operatorname{det}(X) \otimes \operatorname{det}(X)$ is also the trivial representation, $\widehat{H}^{-2}(X,X) \otimes \operatorname{det}(X) \otimes \operatorname{det}(X)$ carries the trivial representation too, proving the lemma in this case.

Now assume that X < Y, so that k > 0. The maximal chains of (X, Y) are of length k-2 and in particular, $C_{k-1}(X, Y) = 0$. So, ker $(\partial_{k-2}^*) = C_{k-2}(X, Y)$ is spanned by the unrefinable chains of the open interval (X, Y). Hence, the map Ψ sending the chain $X_1 < X_2 < \ldots < X_{k-1}$ in $C_{k-2}(X, Y)$ to the element $\varepsilon_X \to \varepsilon_{X_1} \to \varepsilon_{X_2} \to \ldots \to \varepsilon_{X_{k-1}} \to \varepsilon_Y \in \mathbb{C}Q_n$ is a vector space isomorphism from ker (∂_{k-2}^*) to $\varepsilon_Y \mathbb{C}Q_n \varepsilon_X$. By Remark 5.5 and Lemma 5.4, Ψ restricts to a vector space isomorphism from im (∂_{k-3}^*) to ε_Y (ker φ) ε_X .

By definition of the S_n -action on $\mathbb{C}\mathcal{Q}_n$ (Eq. (16)), Ψ is $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ equivariant with respect to a twist by $\det(X) \otimes \det(Y)$. Hence, as $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ -representations,

$$\begin{aligned} (\varepsilon_Y \mathbb{C} \mathcal{Q}_n \varepsilon_X) / (\varepsilon_Y (\ker \varphi) \varepsilon_X) \\ &\cong \left(\ker \left(\partial_{k-2}^* \right) \otimes \det(X) \otimes \det(Y) \right) / \left(\operatorname{im} \left(\partial_{k-3}^* \right) \otimes \det(X) \otimes \det(Y) \right) \\ &\cong \left(\ker \left(\partial_{k-2}^* \right) / \operatorname{im} \left(\partial_{k-3}^* \right) \right) \otimes \det(X) \otimes \det(Y) \\ &\cong \tilde{H}^{k-2} (X, Y) \otimes \det(X) \otimes \det(Y). \end{aligned}$$

The lemma then follows from Eq. (17).

COROLLARY 5.9. Let $\ell(\mu) - \ell(\lambda) = k$. If μ does not refine λ , then $E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu} = 0$. Otherwise, as S_n -representations,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X \leqslant Y]} \left(\tilde{H}_{k-2}\left(X,Y\right) \otimes \det(Y) \otimes \det(X) \right) \Big\uparrow_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}}$$

where the direct sum is over S_n -orbits of pairs $X \leq Y$ in Π_n with $X \in \mu, Y \in \lambda$.

Algebraic Combinatorics, Vol. 8 #2 (2025)

Proof. By Proposition 5.2 and Lemma 5.8, as S_n -representations,

(18)
$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X \leqslant Y]} (E_{Y}\mathbb{C}\mathcal{F}_{n}E_{X}) \Big\uparrow_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}}$$
$$\cong \bigoplus_{[X \leqslant Y]} \tilde{H}^{k-2}(X,Y) \otimes \det(Y) \otimes \det(X) \Big\uparrow_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}}$$

Now, let G be a finite group and let V and W be representations of G over \mathbb{C} . Write * to denote the dual of a representation. Since the vector space isomorphism $V^* \otimes W^* \to (V \otimes W)^*$ sending $f \otimes g$ to the functional $(v \otimes w \mapsto f(v)g(w))$ is G-equivariant, $(V \otimes W)^* \cong V^* \otimes W^*$ as G-representations. Observe that the representation $\det(Y) \otimes \det(X)$ of $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ is self-dual, since it takes values in $\{+1, -1\}$. Hence, as $\operatorname{Stab}_{S_n}(X) \cap \operatorname{Stab}_{S_n}(Y)$ -representations,

$$(\tilde{H}_{k-2}(X,Y) \otimes \det(Y) \otimes \det(X))^* \cong (\tilde{H}_{k-2}(X,Y))^* \otimes (\det(Y) \otimes \det(X))^*$$
$$\cong \tilde{H}^{k-2}(X,Y) \otimes \det(Y) \otimes \det(X).$$

Note also that taking duals commutes with induction (see, for instance Exercise 4.6 of [66].) Thus, continuing Eq. (18),

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{[X\leqslant Y]} \left(\tilde{H}_{k-2}\left(X,Y\right) \otimes \det(Y) \otimes \det(X) \right)^{*} \left(\overset{S_{n}}{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)} \right)^{*}$$
$$\cong \bigoplus_{[X\leqslant Y]} \left(\tilde{H}_{k-2}\left(X,Y\right) \otimes \det(Y) \otimes \det(X) \left(\overset{S_{n}}{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)} \right)^{*} \right)^{*}$$
$$\cong \bigoplus_{[X\leqslant Y]} \tilde{H}_{k-2}\left(X,Y\right) \otimes \det(Y) \otimes \det(X) \left(\overset{S_{n}}{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)} \right)^{*}$$

where the last line follows from the fact that all S_n -representations are self-dual (as they are in fact defined over the rational numbers).

5.3. UPPER INTERVALS. In this section, we fill in the gaps to Step 2 of the proof outline in Section 5.1. In particular, we complete the proof of our "base case" (Proposition 5.1) by proving Lemma 5.18 in Section 5.3.3 and Lemma 5.25 in Section 5.3.5. To do so, we must pin down certain stabilizer representation in Section 5.3.1 and give background on plethysm of formal power series in Section 5.3.2 and necklaces in Section 5.3.4.

5.3.1. *Stabilizer representations.* We study the two representations involved in Eq. (15): the homology of upper intervals in the partition lattice and the determinant representation. The former has been studied by Sundaram in [59]. To explain her result, we first must define some notation.

Let $\lambda = 1^{m_1} 2^{m_2} \dots \vdash n$ be a partition. Given our view of $S_{m_i}[S_i]$ as a subgroup of $S_{m_i i}$ (see Example 2.9), we consider the product of wreath products $\prod_i S_{m_i}[S_i]$ as a subgroup of $S_{m_1+2m_2+3m_3+\dots}$ in the expected manner – see the example below and its accompanying Fig. 2.

EXAMPLE 5.10. Let $\lambda = (4^3, 2^2, 1)$. We consider $S_1 \times S_2[S_2] \times S_3[S_4]$ as a subgroup of S_{17} generated by the following permutations:

- $i = 1, m_i = 1$ (left of three blocks in Fig. 2): None, just the identity.
- $i = 2, m_i = 2$ (middle block in Fig. 2): The two transpositions represented by the arrows within rows (2,3), (4,5) as well as the permutation swapping entire rows q = (2, 4)(3, 5), indicated by the vertical arrow labelled q.

Invariant theory for the face algebra of the braid arrangement



FIGURE 2. $S_1 \times S_2[S_2] \times S_3[S_4]$ as a subgroup of S_{17}

• $i = 4, m_i = 3$ (right block in Fig. 2): The nine transpositions represented by the arrows within rows (6,7), (7,8), (8,9), (10,11), (11,12), (12,13), (14,15), (15,16), (16,17), as well as the two permutations swapping entire rows, represented by the vertical arrows labelled $r_1 = (6,10)(7,11)(8,12)(9,13)$ and $r_2 = (10,14)(11,15)(12,16)(13,17)$.

Given a partition $\lambda = 1^{m_1} 2^{m_2} \dots \vdash n$, define an associated set partition X_{λ} of [n] with $\{1\}, \{2\}, \dots, \{m_1\}$ as its blocks of size one, $\{m_1 + 1, m_1 + 2\}, \{m_1 + 3, m_1 + 4\}, \dots, \{m_1 + 2m_2 - 1, m_1 + 2m_2\}$ as its blocks of size two, and so on. For example,

$$X_{(4^3,2^2,1)} = \{\{1\},\{2,3\},\{4,5\},\{6,7,8,9\},\{10,11,12,13\},\{14,15,16,17\}\}.$$

Note that $\operatorname{Stab}_{S_n}(X_{\lambda}) = \prod_i S_{m_i}[S_i]$. It is clear that the subgroup $\prod_i S_{m_i}[S_i]$ fixes the set partition X_{λ} , and it is routine to check that this is the entire stabilizer subgroup.

Let W_1, W_2, W_3, \ldots be representations of $S_1, S_2, S_3 \ldots$ and let V be a representation of S_r . Assume $m_1 + m_2 + \ldots = r$ and that the restriction $V \downarrow_{\prod_i S_{m_i}}^{S_r}$ decomposes into $\prod_i S_{m_i}$ -irreducibles as

$$\bigoplus_{(\lambda^1,\lambda^2,\ldots)} \left(\chi^{\lambda^1}\otimes\chi^{\lambda^2}\otimes\ldots\right)^{\oplus c(\lambda^1,\lambda^2,\ldots)}$$

Following the notation of Sundaram in [59], define $V \downarrow_{\prod_i S_{m_i}}^{S_r} [W_1 \otimes W_2 \otimes \ldots]$ to be the following representation of $\prod_i S_{m_i}[S_i]$:

$$\bigoplus_{(\lambda^1,\lambda^2,\ldots)} \left(\chi^{\lambda^1}\left[W_1\right] \otimes \chi^{\lambda^2}\left[W_2\right] \otimes \ldots\right)^{\oplus c(\lambda^1,\lambda^2,\ldots)}$$

We are now able to state Sundaram's result on the homology of upper intervals in Π_n .

PROPOSITION 5.11 (Sundaram [59, Proof of Thm. 1.4]). Let $\mu = 1^{m_1} 2^{m_2} \dots \vdash n$ be a partition⁽¹¹⁾ with $\ell(\mu) = r$. Recall that $\hat{1}$ denotes the maximal element $X_{(n)}$ of Π_n . As representations of $\operatorname{Stab}_{S_n}(X_{\mu})$,

$$\tilde{H}_{r-3}\left(X_{\mu},\hat{1}\right) \cong \tilde{H}_{r-3}\left(\Pi_{r}\right) \bigcup_{\prod_{i} S_{m_{i}}}^{S_{r}} [\mathbb{1}_{S_{1}} \otimes \mathbb{1}_{S_{2}} \otimes \ldots].$$

The determinant representation of X_{μ} has a simple formulation.

LEMMA 5.12. Let $\mu = 1^{m_1} 2^{m_2} \dots \vdash n$ be a partition with $\ell(\mu) = r$. As representations of $\operatorname{Stab}_{S_n}(X_{\mu})$,

$$\det(X_{\mu}) \cong \bigotimes_{i \ge 1} \operatorname{sgn}_{S_{m_i}}[\mathbb{1}_{S_i}].$$

⁽¹¹⁾Technically, Sundaram assumes $\mu \neq (n)$ in her proof, but given our convention that $\tilde{H}_{-2}(X, X)$ carries the trivial representation, the proposition holds for $\mu = (n)$ too.

Algebraic Combinatorics, Vol. 8 #2 (2025)

Proof. Let e_1, e_2, \ldots, e_n be standard basis vectors for \mathbb{R}^n . For each block $(X_\mu)_i$ of X_μ , define v_i be the sum

$$v_i \coloneqq \sum_{j \in (X_\mu)_i} e_j.$$

One orientation for the subspace of \mathbb{R}^n associated to X_{μ} is the ordered basis v_1, v_2, \ldots, v_r . Since both $\det(X_{\mu})$ and $\bigotimes_{i \ge 1} \operatorname{sgn}_{S_{m_i}}[\mathbb{1}_{S_i}]$ are one-dimensional characters, it suffices to check that they agree on a generating set for $\operatorname{Stab}_{S_n}(X_{\mu})$. The subgroup $\operatorname{Stab}_{S_n}(X_{\mu}) = \prod_i S_{m_i}[S_i]$ is generated by (i) outer transpositions of the form $(1, 1, \ldots, 1; s_i)$ in one of the copies of $S_{m_j}[S_j]$ and (ii) inner transpositions of the form $(1, \ldots, 1, s_j, 1, \ldots, 1; 1)$ in one of the copies $S_{m_j}[S_j]$. An outer transposition swaps two basis vectors v_i, v_{i+1} , negating the orientation. An inner transposition fixes each v_j , maintaining the orientation. The characters of the generators on $\bigotimes_{i\ge 1} \operatorname{sgn}_{S_{m_i}}[\mathbb{1}_{S_i}]$ agree.

The following lemma will be useful in studying the representation $\tilde{H}_{r-3}(X_{\mu}, \hat{1}) \otimes \det(X_{\mu})$.

LEMMA 5.13. Let V_1, V_2 be representations of S_n and W_1, W_2 be representations of a finite group G. Then, as $S_n[G]$ representations,

$$V_1[W_1] \otimes V_2[W_2] \cong (V_1 \otimes V_2)[W_1 \otimes W_2].$$

Proof. It is straightforward to check that the vector space isomorphism

$$((\otimes^{n} W_{1}) \otimes V_{1}) \otimes ((\otimes^{n} W_{2}) \otimes V_{2}) \longrightarrow (\otimes^{n} (W_{1} \otimes W_{2})) \otimes (V_{1} \otimes V_{2})$$
$$((\otimes_{i} w_{i}^{1}) \otimes v^{1}) \otimes ((\otimes_{i} w_{i}^{2}) \otimes v^{2}) \mapsto (\otimes_{i} (w_{i}^{1} \otimes w_{i}^{2})) \otimes (v^{1} \otimes v^{2}).$$

commutes with the action of $S_n[G]$.

COROLLARY 5.14. Let $\mu = 1^{m_1} 2^{m_2} \dots \vdash n$ be a partition with $\ell(\mu) = r$. As $\operatorname{Stab}_{S_n}(X_{\mu})$ -representations,

$$\tilde{H}_{r-3}(X_{\mu},\hat{1})\otimes \det(X_{\mu})\cong \mathcal{L}_r \bigcup_{\prod_i S_{m_i}}^{S_r} [\mathbb{1}_{S_1}\otimes \mathbb{1}_{S_2}\otimes \ldots].$$

Proof. Using Proposition 5.11 and Lemma 5.12 to rewrite $\hat{H}_{r-3}(X_{\mu}, \hat{1})$ and det (X_{μ}) , we have that as $\operatorname{Stab}_{S_n}(X_{\mu})$ -representations,

$$\tilde{H}_{r-3}\left(X_{\mu},\hat{1}\right)\otimes\det(X_{\mu})\cong\left(\tilde{H}_{r-3}(\Pi_{r})\Big|_{\prod_{i}S_{m_{i}}}^{S_{r}}[\mathbb{1}_{S_{1}}\otimes\mathbb{1}_{S_{2}}\otimes\ldots]\right)\otimes\left(\bigotimes_{i}\operatorname{sgn}_{S_{m_{i}}}[\mathbb{1}_{S_{i}}]\right).$$

By expanding $\tilde{H}_{r-3}(\Pi_r) \downarrow_{\prod_i S_{m_i}}^{S_r} [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots]$, applying Lemma 5.13, then recondensing, the above is equivalent to

$$\left(\tilde{H}_{r-3}(\Pi_r)\Big|_{\prod_i S_{m_i}}^{S_r} \otimes \bigotimes_i \operatorname{sgn}_{S_{m_i}}\right) [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots].$$

However, note that as $\prod_i S_{m_i}$ -representations, $\bigotimes_i \operatorname{sgn}_{S_{m_i}} \cong \operatorname{sgn}_{S_r} \bigcup_{\prod_i S_{m_i}}^{S_r}$. Hence, we can rewrite the above as

Algebraic Combinatorics, Vol. 8 #2 (2025)

$$\left(\tilde{H}_{r-3}(\Pi_r) \Big|_{\prod_i S_{m_i}}^{S_r} \otimes \operatorname{sgn}_{S_r} \Big|_{\prod_i S_{m_i}}^{S_r} \right) [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots]$$

$$\cong \left(\tilde{H}_{r-3}(\Pi_r) \otimes \operatorname{sgn}_{S_r} \right) \Big|_{\prod_i S_{m_i}}^{S_r} [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots]$$

$$\cong \mathcal{L}_r \Big|_{\prod_i S_{m_i}}^{S_r} [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots].$$

5.3.2. *Plethysm of formal power series.* Proving Lemma 5.18 and Lemma 5.25 involves **plethysm of formal power series**, which we explain below using the conventions of Wachs in [65, §2.4].

Let g be a formal power series in the variables $x_1, x_2, \ldots, z_1, z_2, \ldots$ which has nonnegative integer coefficients. Form some sequence $\mathcal{M}(g)$ of the monomials of g, where the number of times each monomial appears in the sequence $\mathcal{M}(g)$ is precisely its coefficient in g. For a symmetric function f in x_1, x_2, \ldots the plethysm f[g] is defined by replacing x_1 in f with the first monomial of $\mathcal{M}(g)$, x_2 with the second monomial of $\mathcal{M}(g)$, and so on. Note that the plethysm f[g] is well-defined because the order of $\mathcal{M}(g)$ does not matter as f is symmetric. We assume that all symmetric functions are written in the variables x_1, x_2, \ldots . If g has no z_i 's and is also symmetric in the x's, the definition of plethysm of formal power series agrees with the ordinary definition of plethysm of f[g].

The following two examples play key roles in the proof of Lemma 5.25.

EXAMPLE 5.15. Let

$$g = \sum_{i \ge 1} z_i h_i = \sum_{\substack{\text{partitions}\\\nu \ne \varnothing}} z_{\ell(\nu)} \mathbf{x}_{\nu}.$$

The monomials in g each have coefficient one. Since there are a countable number of monomials in g there exists some bijection between the monomials and x_1, x_2, \ldots . To compute f[g] for a symmetric function f, one would then replace each x_i in f with its corresponding monomial of g.

EXAMPLE 5.16. Fix a Lyndon word $w = w_1 w_2 \dots w_k$. Let

$$g = \mathbf{z}_w h_w = \prod_{i=1}^k z_{w_i} h_{w_i} = \prod_{i=1}^k z_{w_i} \left(\sum_{\substack{\text{partitions } \nu:\\\ell(\nu) = w_i}} \mathbf{x}_\nu \right).$$

Unlike the previous example, there are some monomials with coefficients higher than one in g. As an example, set w = 12; the monomial $z_1 z_2 x_2 x_3 x_4 = \mathbf{z}_{12} \mathbf{x}_{234}$ has coefficient three in $g = (z_1 h_1)(z_2 h_2)$, counting which of the x_2 , x_3 , or x_4 arise from h_1 . To distinguish between repeated monomials, we consider instead distinct ordered tuples

$$(z_{w_1}\mathbf{x}_{\nu(1)}, z_{w_2}\mathbf{x}_{\nu(2)}, \ldots, z_{w_k}\mathbf{x}_{\nu(k)}),$$

where each $\nu(i)$ is a partition with length w_i . For instance, the three ordered pairs producing $\mathbf{z}_{12}\mathbf{x}_{234}$ are

$$(z_1\mathbf{x}_2, z_2\mathbf{x}_{34}), (z_1\mathbf{x}_3, z_2\mathbf{x}_{24}), \text{ and } (z_1\mathbf{x}_4, z_2\mathbf{x}_{23}).$$

There are countably many such ordered pairs so there exists some bijection matching the ordered pairs to x_1, x_2, \ldots . To compute f[g] for a symmetric function f, one would then replace each x_i in f with the monomial of g associated to the corresponding ordered pair.

Algebraic Combinatorics, Vol. 8 #2 (2025)

5.3.3. Proof of Lemma 5.18. To manipulate the generating function $\sum_{\mu\neq 0} \mathbf{z}_{\mu} \cdot ch\left(E_{|\mu|}\mathbb{C}\mathcal{F}_{|\mu|}E_{\mu}\right)$, we will need a technical but powerful lemma. The following lemma is an application of a formula of Wachs from [64, Theorem 5.5]⁽¹²⁾, which is a refinement of a result of Sundaram in [59, Lemma 1.5].

LEMMA 5.17 (Wachs [64], refining Sundaram [59]). For each positive integer i, pick two S_i -modules: V_i and W_i . Then,

$$\sum_{r \ge 1} \sum_{\substack{\mu = \\ 1^{m_1 2^{m_2} \dots : \\ m_1 + m_2 + \dots = r}}} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(V_r \bigcup_{\prod_{i \ge 1} S_{m_i}}^{S_r} [W_1 \otimes W_2 \otimes \dots] \right)^{\sum_{i \ge 1} S_{m_i}[S_i]} \right)$$
$$= \sum_{r \ge 1} \operatorname{ch} \left(V_r \right) \left[\sum_{i \ge 1} z_i \cdot \operatorname{ch} \left(W_i \right) \right].$$

With this lemma, we are now ready to prove Lemma 5.18.

Lemma 5.18.

$$\sum_{\mu \neq \varnothing} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(E_{|\mu|} \mathbb{C} \mathcal{F}_{|\mu|} E_{\mu} \right) = \sum_{r \ge 1} L_r [z_1 h_1 + z_2 h_2 + \ldots].$$

Proof. Let $\mu \neq \emptyset$ be a partition of n of the form $1^{m_1}2^{m_2}\dots$ with $\ell(\mu) = r$. By Corollary 5.9,

$$\operatorname{ch}\left(E_{|\mu|}\mathbb{C}\mathcal{F}_{n}E_{\mu}\right) = \operatorname{ch}\left(\tilde{H}_{r-3}\left(X_{\mu},\hat{1}\right)\otimes\operatorname{det}(X_{\mu})\right)^{S_{n}}_{\prod_{i}S_{m_{i}}[S_{i}]}\right).$$

By Corollary 5.14,

$$\tilde{H}_{r-3}(X_{\mu},\hat{1})\otimes \det(X_{\mu})\cong \mathcal{L}_r \Big|_{\prod_{i\geq 1}S_{m_i}}^{S_r}[\mathbb{1}_{S_1}\otimes\mathbb{1}_{S_2}\otimes\ldots].$$

Therefore,

$$\operatorname{ch}(E_{|\mu|}\mathbb{C}\mathcal{F}_{|\mu|}E_{\mu}) = \operatorname{ch}\left(\left(\mathcal{L}_r \bigcup_{\substack{S_r \\ \prod_{i \ge 1} S_{m_i}}} [\mathbb{1}_{S_1} \otimes \mathbb{1}_{S_2} \otimes \ldots]\right) \Big|_{\prod_{i \ge 1} S_{m_i}[S_i]}^{S_n}\right).$$

Substituting $V_i = \mathcal{L}_i$ and $W_i = \mathbb{1}_{S_i}$ into Lemma 5.17, we have that

$$\sum_{\mu \neq \varnothing} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(E_{|\mu|} \mathbb{C} \mathcal{F}_{|\mu|} E_{\mu} \right) = \sum_{r \geq 1} \sum_{\substack{\mu = \\ 1^{m_{1}} 2^{m_{2}} \dots \\ m_{1} + m_{2} + \dots = r}} \mathbf{z}_{\mu} \cdot \operatorname{ch} \left(E_{|\mu|} \mathbb{C} \mathcal{F}_{|\mu|} E_{\mu} \right)$$
$$= \sum_{r \geq 1} L_{r} [z_{1}h_{1} + z_{2}h_{2} + \dots].$$

5.3.4. Necklaces. Let \mathcal{A} be a totally ordered alphabet. A **necklace** of length $k \ge 1$ on \mathcal{A} is an equivalence class of a word on \mathcal{A} with k letters up to cyclic rotation. We will write the necklace represented by the word (w_1, w_2, \ldots, w_k) as $[w_1, w_2, \ldots, w_k]$. Define $\mathcal{N}_k(\mathcal{A})$ to be the set of necklaces on \mathcal{A} with k letters, and $\mathcal{N}(\mathcal{A})$ to be the set of necklaces on \mathcal{A} of any (nonzero, finite) length.

 $^{^{(12)}}$ Wachs uses different notation here.

Algebraic Combinatorics, Vol. 8 #2 (2025)

DEFINITION 5.19. Let $\mathcal{A} = \{1, 2, ...\}$ be the standard ordered alphabet. We define a map

$$f: \mathcal{N}(\mathcal{A}) \longrightarrow \bigsqcup_{Lyndon} \{(w, 1), (w, 2), (w, 3), \ldots\}$$

as follows. Let η be a necklace on \mathcal{A} with a lexicographically minimal representative word $\alpha = (\alpha_1, \alpha_2, \ldots, \alpha_k)$. Notice that α is either a Lyndon word, or a power $i \ge 1$ of a single Lyndon word, so can be written as $\alpha = w^i$ for a Lyndon word w. The Lyndon word w and power i are unique since Lyndon factorization is unique. Let fbe the map which sends η to the pair (w, i).

Although we won't need this fact, the map f is a bijection since $(w, i) \mapsto [w^i]$ can be checked to be an inverse (using that the w are Lyndon words).

EXAMPLE 5.20. f([2, 2, 1, 1, 2, 2, 1, 1, 2, 2, 1, 1]) = ((1, 1, 2, 2), 3).

A necklace is **primitive** if its representative words are not powers of shorter words. For example, the necklace [2, 2, 1, 1, 2, 2, 1, 1, 2, 2, 1, 1] in Example 5.20 is *not* primitive since it is represented by the word $(2, 2, 1, 1)^3$. Let $\mathcal{PN}_k(\mathcal{A})$ denote the set of primitive necklaces of length k on the alphabet \mathcal{A} and let $\mathcal{PN}(\mathcal{A})$ denote the set of primitive necklaces on \mathcal{A} of any (nonempty, finite) length.

Recall the notion of a sequence $\mathcal{M}(g)$ associated to a formal power series g from Section 5.3.2. If g has a countable number of monomials, we can theoretically view $\mathcal{M}(g)$ as a totally ordered alphabet (although we won't need to refer to the order) and form words and necklaces on it. Given a word $w = (\mathbf{z}_{\nu(1)}\mathbf{x}_{\mu(1)}, \mathbf{z}_{\nu(2)}\mathbf{x}_{\mu(2)}, \dots, \mathbf{z}_{\nu(k)}\mathbf{x}_{\mu(k)})$ on $\mathcal{M}(g)$ (where each $\nu(i), \mu(i)$ are partitions), define its **evaluation** eval(w) to be the product

$$\operatorname{eval}(w) \coloneqq \prod_{i=1}^{k} \mathbf{z}_{\nu(i)} \mathbf{x}_{\mu(i)}.$$

Since evaluation does not depend on the order of the letters in w, we can define it on necklaces in $\mathcal{N}(\mathcal{M}(g))$, by $\operatorname{eval}([w]) = \operatorname{eval}(w)$.

EXAMPLE 5.21. Recall the examples in Section 5.3.2. Let η and τ be the following necklaces in $\mathcal{PN}(\mathcal{M}(z_1h_1 + z_2h_2 + \ldots))$ and $\mathcal{PN}(\mathcal{M}(\mathbf{z}_{223}h_{223}))$, respectively. Note that we are using our ordered pair notation for the necklaces on $\mathcal{M}(\mathbf{z}_{223}h_{223})$.

 $\eta = \left[z_3 \mathbf{x}_{244}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{266}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{246}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}\right],$

 $\tau = \left[\left(z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{266} \right), \left(z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{246} \right), \left(z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{244} \right) \right].$

Then,

$$eval(\eta) = eval(\tau) = \mathbf{z}_{2^{6}3^{3}}\mathbf{x}_{2^{6}3^{3}4^{9}6^{3}} = z_{2}^{6}z_{3}^{3}x_{2}^{6}x_{3}^{3}x_{4}^{9}x_{6}^{3}.$$

5.3.5. *Necklace bijection*. We define an evaluation-preserving map

$$\Psi: \mathfrak{PN}\left(\mathcal{M}\left(z_1h_1+z_2h_2+\ldots\right)\right) \longrightarrow \bigsqcup_{\substack{\text{Lyndon}\\w}} \mathfrak{PN}\left(\mathcal{M}\left(\mathbf{z}_wh_w\right)\right).$$

Consider the necklace

$$\eta = \left[z_{\ell(\nu(1))} \mathbf{x}_{\nu(1)}, z_{\ell(\nu(2))} \mathbf{x}_{\nu(2)}, \dots, z_{\ell(\nu(k))} \mathbf{x}_{\nu(k)} \right]$$

in $\mathcal{PN}(\mathcal{M}(z_1h_1 + z_2h_2 + \ldots))$. Define an associated necklace $z(\eta)$ on $\{1, 2, \ldots\}$ by taking the subscripts of z in η . Specifically, define

$$z(\eta) \coloneqq \left[\ell\left(\nu(1)\right), \ell\left(\nu(2)\right), \dots, \ell\left(\nu(k)\right)\right].$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

Recall the map f in Definition 5.19; we have $f([z(\eta)]) = (w, k/i)$ for some Lyndon word w and integer i, with $[z(\eta)] = [w^{k/i}]$. Because w is a Lyndon word, if we write \overline{m} to mean m modulo k, there are precisely k/i values of $0 \leq j \leq k-1$ such that

(19)
$$w = (w_1, w_2, \dots, w_i) = \left(\ell\left(\nu(\overline{j+1})\right), \ell\left(\nu(\overline{j+2})\right), \dots, \ell\left(\nu(\overline{j+i})\right)\right).$$

In words, the map Ψ will (i) pick one of the k/i representative words of η whose prefix has w_1, w_2, \ldots, w_i as the subscripts of z then (ii) group together the first *i* letters, the next *i* letters, and so on. The result is a word on $\mathcal{M}(\mathbf{z}_w h_w)$ and Ψ sends η to the necklace represented by that word. More formally, $\Psi(\eta)$ is the following necklace on $\mathcal{M}(\mathbf{z}_w h_w)$ with k/i beads:

$$\begin{split} \Psi(\eta) \coloneqq \left[\left(z_{w_1} \mathbf{x}_{\nu(\overline{j+1})}, z_{w_2} \mathbf{x}_{\nu(\overline{j+2})}, \dots, z_{w_i} \mathbf{x}_{\nu(\overline{j+i})} \right), \\ \left(z_{w_1} \mathbf{x}_{\nu(\overline{j+i+1})}, z_{w_2} \mathbf{x}_{\nu(\overline{j+i+2})}, \dots, z_{w_i} \mathbf{x}_{\nu(\overline{j+2i})} \right), \\ \dots, \\ \left(z_{w_1} \mathbf{x}_{\nu(\overline{j+k-i+1})}, z_{w_2} \mathbf{x}_{\nu(\overline{j+k-i+2})}, \dots, z_{w_i} \mathbf{x}_{\nu(\overline{j+k})} \right) \right]. \end{split}$$

The map Ψ is well-defined because the resulting necklace does not depend on the choice of j from Eq. (19). The necklace $\Psi(\eta)$ is primitive because η is, and Ψ is evaluation preserving by construction.

EXAMPLE 5.22. Let η be the following primitive necklace in $\mathcal{PN}(\mathcal{M}(z_1h_1+z_2h_2+\ldots))$:

 $\eta = [z_3 \mathbf{x}_{244}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{266}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}, z_3 \mathbf{x}_{246}, z_2 \mathbf{x}_{44}, z_2 \mathbf{x}_{23}].$

The necklace $z(\eta)$ is [3, 2, 2, 3, 2, 2, 3, 2, 2], which f maps to the ordered pair ((2, 2, 3), 3). The possible choices of beads which we can cyclically rotate to the front before grouping are colored blue. $\Psi(\eta)$ is the following primitive necklace on $\mathcal{PN}(\mathcal{M}(\mathbf{z}_{223}h_{223}))$:

 $\left[\left(z_{2}\mathbf{x}_{44}, z_{2}\mathbf{x}_{23}, z_{3}\mathbf{x}_{266}\right), \left(z_{2}\mathbf{x}_{44}, z_{2}\mathbf{x}_{23}, z_{3}\mathbf{x}_{246}\right), \left(z_{2}\mathbf{x}_{44}, z_{2}\mathbf{x}_{23}, z_{3}\mathbf{x}_{244}\right)\right].$

As explained in Example 5.21, η and $\Psi(\eta)$ have the same evaluation.

PROPOSITION 5.23. The map Ψ is a bijection.

Proof. We prove that Ψ is a bijection by providing an inverse map

$$\Theta: \bigsqcup_{\substack{\text{Lyndon}\\w}} \mathcal{PN}\left(\mathcal{M}\left(\mathbf{z}_w h_w\right)\right) \longrightarrow \mathcal{PN}\left(\mathcal{M}\left(z_1 h_1 + z_2 h_2 + \ldots\right)\right).$$

Let $w = (w_1, w_2, \ldots, w_\ell)$ be a Lyndon word and let τ be the following primitive necklace in $\mathcal{PN}(\mathcal{M}(\mathbf{z}_w h_w))$:

 $\tau = \left[\left(z_{w_1} \mathbf{x}_{\nu(1,1)}, z_{w_2} \mathbf{x}_{\nu(1,2)}, \dots, z_{w_\ell} \mathbf{x}_{\nu(1,\ell)} \right), \dots, \left(z_{w_1} \mathbf{x}_{\nu(m,1)}, z_{w_2} \mathbf{x}_{\nu(m,2)}, \dots, z_{w_\ell} \mathbf{x}_{\nu(m,\ell)} \right) \right].$

Define Θ to be the map which removes parentheses, sending τ to the following necklace on $\mathcal{M}(z_1h_1 + z_2h_2 + \ldots)$,

 $\begin{bmatrix} z_{w_1}\mathbf{x}_{\nu(1,1)}, z_{w_2}\mathbf{x}_{\nu(1,2)}, \dots, z_{w_\ell}\mathbf{x}_{\nu(1,\ell)}, z_{w_1}\mathbf{x}_{\nu(2,1)}, \dots, z_{w_\ell}\mathbf{x}_{\nu(2,\ell)}, \dots, z_{w_\ell}\mathbf{x}_{\nu(m,\ell)} \end{bmatrix}.$ We claim $\Theta(\tau)$ is a primitive necklace. If there were a prefix $z_{w_1}\mathbf{x}_{\nu(1,1)}\dots z_{w_i}\mathbf{x}_{\nu(j,i)}$ with $i < \ell$ and power k for which

$$\left(z_{w_1}\mathbf{x}_{\nu(1,1)}\ldots z_{w_i}\mathbf{x}_{\nu(j,i)}\right)^k = z_{w_1}\mathbf{x}_{\nu(1,1)}\ldots z_{w_\ell}\mathbf{x}_{\nu(m,\ell)},$$

then by restricting our attention to the coefficients of z, we'd see $(w^{j-1}w_1\cdots w_i)^k = w^m$, which implies that $w_1w_2\cdots w_{\ell-i} = w_{i+1}\cdots w_\ell$. This is not possible since w is

Algebraic Combinatorics, Vol. 8 #2 (2025)

Lyndon, and an equivalent characterization of w being Lyndon is that it is strictly lexicographically smaller than all of its proper suffixes (see [36, Proposition 5.12]). There are also no factorizations of the form

$$\left(z_{w_1}\mathbf{x}_{\nu(1,1)}\dots z_{w_\ell}\mathbf{x}_{\nu(j,\ell)}\right)^k = z_{w_1}\mathbf{x}_{\nu(1,1)}\dots z_{w_\ell}\mathbf{x}_{\nu(m,\ell)}$$

since τ is primitive. Hence, $\Theta(\tau)$ is a primitive necklace of $\mathcal{M}(z_1h_1 + z_2h_2 + \cdots)$. It is straightforward to check that $\Theta \circ \Psi$ is the identity mapping. The fact that w is Lyndon implies $z(\Theta(\tau))$ is necessarily w^m , hence $\Psi \circ \Theta$ is the identity mapping as well, completing the proof.

The symmetric function L_n , and in fact all the higher Lie symmetric functions L_{λ} , have combinatorial interpretations involving primitive necklaces (see Gessel–Reutenauer's⁽¹³⁾ work: [26, Equation 2.1]). The next lemma is an application of this interpretation.

LEMMA 5.24. Let g be a formal power series in $x_1, x_2, \ldots, z_1, z_2, \ldots$ which has nonnegative integer coefficients. Then,

$$\sum_{n \ge 1} L_n[g] = \sum_{\eta \in \mathcal{PN}(\mathcal{M}(g))} \operatorname{eval}(\eta).$$

Proof. In [26, Equation 2.1] (by setting $\lambda = n$), Gessel–Reutenauer explain that

$$L_n = \sum_{\tau \in \mathcal{PN}_n(x_1, x_2, \ldots)} \operatorname{eval}(\tau).$$

The lemma then follows from the definition of plethysm of formal power series (Section 5.3.2). $\hfill \Box$

We are finally ready to prove Lemma 5.25.

Lemma 5.25.

$$\sum_{r \ge 1} L_r[z_1h_1 + z_2h_2 + \ldots] = \sum_{\substack{Lyndon \\ w}} \sum_{m \ge 1} \mathbf{z}_{w^m} L_m[h_w].$$

Proof. By Proposition 5.23, there is an evaluation-preserving bijection between

$$\bigsqcup_{\substack{\text{Lyndon}\\w}} \mathcal{PN}\left(\mathcal{M}\left(\mathbf{z}_w h_w\right)\right) \text{ and } \mathcal{PN}\left(\mathcal{M}\left(z_1 h_1 + z_2 h_2 + \ldots\right)\right).$$

Hence, we have the equality

$$\sum_{\substack{\text{Lyndon}\\m}} \sum_{\eta \in \mathcal{PN}(\mathcal{M}(\mathbf{z}_w h_w))} \operatorname{eval}(\eta) = \sum_{\eta \in \mathcal{PN}(\mathcal{M}(z_1 h_1 + \ldots))} \operatorname{eval}(\eta).$$

Thus, by Lemma 5.24,

$$\sum_{\substack{\text{Lyndon} \\ w}} \sum_{m \ge 1} L_m[\mathbf{z}_w h_w] = \sum_{r \ge 1} L_r[z_1 h_1 + z_2 h_2 + \ldots].$$

The lemma then follows from the observation that

$$L_m[\mathbf{z}_w h_w] = \mathbf{z}_{w^m} L_m[h_w].$$

 $^{(13)}$ Gessel–Reutenauer used this necklace interpretation to prove their remarkable theorem mentioned in the proof of Proposition 4.6.

Algebraic Combinatorics, Vol. 8 #2 (2025)

 \Box

5.4. GENERAL INTERVALS. In this section, we complete the proofs from Step 3 of the proof outline in Section 5.1. In Section 5.4.1, we combinatorially interpret the S_n -orbits of $X \leq Y$ and identify their stabilizer subgroups. In Section 5.4.2, we compute the representations on the homology of intervals of Π_n and their associated determinant twists, allowing us to understand the representations $E_Y \mathbb{CF}_n E_X$ in terms of Step 2.

5.4.1. Identifying intersections of stabilizers.

DEFINITION 5.26. In a **multiset partition**, repeated elements are viewed as indistinguishable. For example, there are precisely four multiset partitions of the multiset $\{1, 1, 2\}$: $\{112\}, \{1, 12\}, \{11, 2\}, \{1, 1, 2\}$. Given partitions λ, μ of the same size we define a set $\lambda(\mu)$ of multiset partitions. In particular, if μ does not refine λ , set $\lambda(\mu)$ to be the empty set. Otherwise, define $\lambda(\mu)$ to be the set of multiset partitions of the multiset $\{\mu_i : 1 \leq i \leq \ell(\mu)\}$ which have block sizes summing to the parts of λ .

EXAMPLE 5.27. • Let $\mu = (5, 3, 2, 2, 1, 1)$ and $\lambda = (6, 5, 3)$. Then,

 $\lambda(\mu) = \{\{2211, 5, 3\}, \{51, 32, 21\}, \{51, 221, 3\}, \{321, 5, 21\}\}.$

• Let $\mu = (2, 2, 1, 1, 1, 1)$ and $\lambda = (3, 3, 2)$. Then,

 $\lambda(\mu) = \{\{21, 21, 11\}, \{21, 111, 2\}\}.$

LEMMA 5.28. There is a bijection

 $\{S_n \text{-}orbits \ [X \leq Y] : X \in \mu, Y \in \lambda\} \longleftrightarrow \lambda(\mu).$

Proof. Consider the map which sends $[X \leq Y]$ to the multiset partition in $\lambda(\mu)$ which has one block ν for each block Y_i of Y, where ν is comprised of the sizes $\nu_1, \nu_2, \ldots, \nu_{\ell(\nu)}$ of the blocks in $X_{j_1}, \ldots, X_{j_{\ell(\nu)}}$ which combine to form Y_i . To see that this map is well-defined, observe that for $w \in S_n$, the blocks of w(X) which form $w(Y_i)$ are $w(X_{j_1}), \ldots, w(X_{j_{\ell(\nu)}})$, whose sizes are preserved.

Now, we prove the map is a bijection. First, note that each orbit has a representative of the form $X_{\mu} \leq Y$, since S_n acts transitively on X with block sizes μ . Fixing a multiset partition S in $\lambda(\mu)$, one can always build a set partition Y with block sizes λ for which $[X_{\mu} \leq Y]$ maps to S by combining blocks of the appropriate sizes in X_{μ} to form Y. Hence, the map is surjective. To see injectivity, assume $[X_{\mu} \leq Y]$ and $[X_{\mu} \leq Y']$ map to the same multiset partition in $\lambda(\mu)$. By construction of the map, one can then make permutations swapping *entire blocks* of the same size in X_{μ} to get from Y to Y'. A permutation formed from such swaps is necessarily in $\operatorname{Stab}_{S_n}(X_{\mu})$, so $[X_{\mu} \leq Y] = [X_{\mu} \leq Y']$, as desired. \Box

It will be useful for us to specify a canonical representative of each orbit $[X \leq Y]$; equivalently, given an $S \in \lambda(\mu)$, we specify a canonical $X_S \leq Y_S$ for which $[X_S \leq Y_S]$ maps to S.

DEFINITION 5.29 (Definition of X_S, Y_S). Fix a multiset partition $S \in \lambda(\mu)$. Define $Y_S := X_{\lambda}$. Place an ordering on the blocks of $Y_S = Y_1, Y_2, \ldots, Y_{\ell(\lambda)}$ so that the minimum elements of the blocks increase. We place an ordering also on the blocks ν of $S = \{\nu^{m_{\nu}}\}$: first by size of $|\nu|$, then breaking ties by increasing in lexicographical order (when the parts of ν are ordered to weakly decrease). In this way, each block B in Y_S is labelled by a partition $\nu \in S$ with $\nu \vdash \#B$. To form the set partition X_S from Y_S , partition each block B of Y_S as follows. If the partition labelling B is $\nu = 1^{m_1}2^{m_2}\ldots$, then let the m_1 smallest numbers of B be blocks of size one, let the next $2m_2$ smallest numbers of B be blocks of size 2 (grouping together the smallest two, then the next smallest two, etc) and so on.

Algebraic Combinatorics, Vol. 8 #2 (2025)

EXAMPLE 5.30. As an example, consider $S = \{3, 3, 3, 211, 211, 22, 22\} \in \lambda(\mu)$, for $\lambda = 4^4 3^3$ and $\mu = 3^3 2^6 1^4$. The partitions $\nu \in S$ are already written in the order described in Definition 5.29. Y_S (with its associated labels by the $\nu \in S$) is

$$Y_{S} = \left\{ \underbrace{\{1, 2, 3\}}_{\nu=3}, \underbrace{\{4, 5, 6\}}_{\nu=3}, \underbrace{\{7, 8, 9\}}_{\nu=3}, \underbrace{\{10, 11, 12, 13\}}_{\nu=211}, \underbrace{\{14, 15, 16, 17\}}_{\nu=211}, \underbrace{\{18, 19, 20, 21\}}_{\nu=22}, \underbrace{\{22, 23, 24, 25\}}_{\nu=22} \right\}.$$

Using the algorithm from Definition 5.29, we obtain X_S as

$$\begin{split} X_S = & \Big\{\{1,2,3\},\{4,5,6\},\{7,8,9\},\{10\},\{11\},\{12,13\},\{14\},\{15\},\{16,17\},\{18,19\},\\ & \{20,21\},\{22,23\},\{24,25\}\Big\}. \end{split}$$

For a fixed multiset partition $S = \{\nu^{m_{\nu}} : \nu \in S\} \in \lambda(\mu)$ with $|\lambda| = n$, consider the subgroup of S_n generated by following two types of permutations:

- **Outer:** For each $\nu \in S$: the permutations which swap two entire blocks of Y_S labelled by the same partition ν , and leave the remaining integers alone i.e. if Y_i, Y_j are two blocks labelled by ν , the permutation which swaps the smallest numbers of Y_i, Y_j , then the second smallest numbers, and so on, but fixes all $k \notin Y_i \cup Y_j$.)
- Inner: For each block B of Y_S , any permutation which maintains the set partition of B given by its subblocks in X_S (and fixes all $i \notin B$).

The union of these types of permutations generate a subgroup of S_n isomorphic to

$$\prod_{\nu \in S} S_{m_{\nu}} \left[\operatorname{Stab}_{S_{|\nu|}} (X_{\nu}) \right]$$

In fact, this shall be our definition of how to view $\prod_{\nu \in S} S_{m_{\nu}} [\operatorname{Stab}_{S|\nu|}(X_{\nu})]$ as a subgroup of S_n . Writing $\lambda \vdash n$ as $1^{m_1}2^{m_2}\ldots$, note that $\prod_{\nu \in S} S_{m_{\nu}} [\operatorname{Stab}_{S|\nu|}(X_{\nu})]$ embeds naturally into $\prod_{\nu} S_{m_{\nu}} [S_{|\nu|}]$, which in turn embeds naturally into $S_{m_1,2m_2,3m_3,\ldots}$ and then into S_n .

EXAMPLE 5.31. We point out how to view $S_3[S_3] \times S_2[S_2[S_1] \times S_1[S_2]] \times S_2[S_2[S_2[S_2]]$ as a subgroup of S_{25} pictorially in Fig. 3. The twenty generators are indicated by the twenty arrows. Given the similarities to Example 2.9 and Example 5.10, we do not write out all the generators here. However, we note that the twelve horizontal arrows within rows indicate adjacent transpositions, the six vertical arrows between rows indicate swaps of *entire rows* – these eighteen total arrows generate the *inner* permutations. The two labelled vertical arrows represent swaps of *entire blocks*. In particular, q = (10, 14)(11, 15)(12, 16)(13, 17) and r = (18, 22)(19, 23)(20, 24)(21, 25)generate the *outer* permutations.

LEMMA 5.32. Let $S = \{\nu^{m_{\nu}} : \text{partition } \nu\}$ be a multiset partition in $\lambda(\mu)$. Then,

$$\operatorname{Stab}_{S_n}(X_S) \cap \operatorname{Stab}_{S_n}(Y_S) = \prod_{\nu \in S} S_{m_\nu} \left[\operatorname{Stab}_{S_{|\nu|}}(X_{\nu}) \right].$$

Proof. By construction, every generator of $\prod_{\nu \in S} S_{m_{\nu}} [\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ fixes both Y_S and X_S , and so the same is true for every element. Hence, $\prod_{\nu \in S} S_{m_{\nu}} [\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ is a subgroup of $\operatorname{Stab}_{S_n}(X_S) \cap \operatorname{Stab}_{S_n}(Y_S)$. To prove equality of the subgroups, we shall show their sizes are the same.

Algebraic Combinatorics, Vol. 8 #2 (2025)



FIGURE 3. $S_3[S_3] \times S_2[S_2[S_1] \times S_1[S_2]] \times S_2[S_2[S_2]]$ as a subgroup of S_{25} .

Observe that $\operatorname{Stab}_{S_n}(X_S \leq Y_S) = \operatorname{Stab}_{S_n}(X_S) \cap \operatorname{Stab}_{S_n}(Y_S)$. By the orbit-stabilizer theorem, it suffices to prove

(20)
$$|[X_S \leqslant Y_S]| = \frac{n!}{\left| \prod_{\nu \in S} S_{m_{\nu}} \left[\text{Stab}_{S_{|\nu|}} \left(X_{\nu} \right) \right] \right|}.$$

Given a partition $\nu = 1^{n_1} 2^{n_2} \dots$, define a constant $N_{\nu} = \prod_i n_i ! (i!)^{n_i}$. Observe that $|\operatorname{Stab}_{S|\nu|}(X_{\nu})| = N_{\nu}$, so that the right side of Eq. (20) can be rewritten as

$$\frac{n!}{\left|\prod_{\nu\in S} S_{m_{\nu}}\left[\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})\right]\right|} = \frac{n!}{\prod_{\nu\in S} m_{\nu}! \left(N_{\nu}\right)^{m_{\nu}}}.$$

The left side of Eq. (20) counts the number of distinct pairs $X \leq Y \in \Pi_n$ so that $[X \leq Y]$ maps to S. The number of ways to pick a set partition $Y \in \lambda$ is $\frac{n!}{N_{\lambda}}$. To pick a set partition X refining Y so that $[X \leq Y]$ maps to S, we must (i) pick which m_{ν} blocks of Y of size *i* get assigned to be partitioned into blocks of sizes for each $\nu \in S$ with $\nu \vdash i$ then (ii) count the number of ways to break up each block of type ν .

Let $m_i(\lambda)$ denote the number of parts of size *i* within the partition λ . The number of ways to assign the blocks of size *i* the partitions $\nu \vdash i$ is $\frac{m_i(\lambda)!}{\prod_{\nu \vdash i} m_{\nu}!}$. The number of ways to break up a block of size *i* into blocks with sizes the parts of ν is $\frac{i!}{N_{\nu}}$. Putting it all together, the left side of Eq. (20) is

$$\frac{n!}{N_{\lambda}} \prod_{i} \left(\frac{m_{i}(\lambda)!}{\prod_{\nu \vdash i} m_{\nu}!} \prod_{\nu \vdash i} \left(\frac{i!}{N_{\nu}} \right)^{m_{\nu}} \right) = n! \prod_{i} \left(\frac{1}{\prod_{\nu \vdash i} m_{\nu}!} \prod_{\nu \vdash i} \left(\frac{1}{N_{\nu}} \right)^{m_{\nu}} \right)$$
$$= \frac{n!}{\prod_{\nu} m_{\nu}! \left(N_{\nu} \right)^{m_{\nu}}},$$

as desired.

5.4.2. Twisted homology of intervals in the partition lattice.

LEMMA 5.33. Let $S = \{\nu^{m_{\nu}} : \text{partition } \nu\}$ be a multiset partition in $\lambda(\mu)$. As representations of $\prod_{\nu \in S} S_{m_{\nu}}[\operatorname{Stab}_{S|\nu|}(X_{\nu})],$

$$\det(X_S) \otimes \det(Y_S) \cong \bigotimes_{\nu \in S} \rho(\nu) \left[\det(X_{\nu})\right],$$

where $\rho(\nu) = \mathbb{1}_{S_{m_{\nu}}}$ if $\ell(\nu)$ is odd and $\rho(\nu) = \operatorname{sgn}_{S_{m_{\nu}}}$ if $\ell(\nu)$ is even.

Proof. First, we claim that as $\prod_{\nu} S_{m_{\nu}} [\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ -representations,

(21)
$$\det(X_S) \cong \bigotimes_{\nu \in S} \left(\rho(\nu) \otimes \operatorname{sgn}_{S_{m_{\nu}}} \right) \left[\det(X_{\nu}) \right],$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

458

Since both are one-dimensional characters, it suffices to check they agree on a generating set of $\prod_{\nu} S_{m_{\nu}} [\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$. Order the blocks B_1, B_2, \ldots, B_k of X_S so that the minimal element of each block increases. An ordered basis for the space associated to X_S is v_1, v_2, \ldots, v_k , where v_i is the sum of the standard vectors e_i for the integers j appearing in the block B_i of X_S .

Recall the notion of outer versus inner generators from the discussion preceding Example 5.31. An *inner* generator w associated to a block of Y_S labelled by ν formed from the blocks $B_{i+1}, \ldots, B_{i+\ell(\nu)}$ of X acts block diagonally on v_1, v_2, \ldots, v_k . In particular, it acts trivially on every basis vector except for the block $v_{i+1}, \ldots, v_{i+\ell(\nu)}$, which it permutes amongst itself. The submatrix corresponding to $v_{i+1}, \ldots, v_{i+\ell(\nu)}$ has determinant $\det(X_{\nu})(w)$. An outer transposition associated to ν swaps $\ell(\nu)$ basis vectors with another $\ell(\nu)$ basis vectors – this is represented by a matrix with determinant +1 if $\ell(\nu)$ is even and -1 if $\ell(\nu)$ is odd. Hence, Eq. (21) holds.

We now show that as $\prod_{\nu} S_{m_{\nu}} [\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ -representations,

(22)
$$\det(Y_S) \cong \bigotimes_{\nu \in S} \operatorname{sgn}_{S_{m_{\nu}}} \left[\mathbb{1}_{S_{|\nu|}} \right].$$

Again, it suffices to check on the generators. If C_1, C_2, \ldots, C_r are the blocks of Y_S , one ordered basis for Y_S is u_1, u_2, \ldots, u_r , where u_i is the sum of the v_j for which $B_j \subseteq C_i$. The inner generators act trivially on this ordered basis and an outer generator swaps u_i, u_j for two blocks C_i, C_j of Y_S indexed by the same partition, hence is orientation reversing. Thus, Eq. (22) holds and the lemma then follows from Lemma 5.13.

To compute the homology of intervals in the partition lattice, we will use the extremely useful tools Sundaram explains in [58]. If P and Q are posets, let $P \times Q$ denote the **Cartesian product** of P and Q, which is the set $P \times Q$ under the relation $(p,q) \leq (p',q')$ if and only if $p \leq p'$ in P and $q \leq q'$ in Q. We write P^n to denote the *n*-fold Cartesian product $\underbrace{P \times P \times \ldots \times P}_{n \text{ times}}$.

PROPOSITION 5.34 (Sundaram, [58]). Let P and Q be posets, each with a minimal element $\hat{0}$ and a maximal element $\hat{1}$. Let G_P , G_Q be groups acting by poset automorphisms on P and Q respectively. Assume that for $i \neq r$, the homology groups $\tilde{H}_i(P) = 0$, and that for $j \neq s$, the homology groups $\tilde{H}_j(Q) = 0$.

(i) [58, Proposition 2.1]: $\tilde{H}_i(P \times Q) = 0$ unless i = r + s + 2, and as $G_P \times G_Q$ representations,

$$H_{r+s+2}(P \times Q) \cong H_r(P) \otimes H_s(Q).$$

(ii) [58, Proposition 2.3]⁽¹⁴⁾: $\tilde{H}_i(P^n) = 0$ unless i = n(r+2) - 2 and as $S_n[G_P]$ representations,

$$\tilde{H}_{n(r+2)-2}(P^n) \cong \begin{cases} \mathbb{1}_{S_n} \left[\tilde{H}_r(P) \right] & \text{if } r \text{ is even,} \\ \operatorname{sgn}_{S_n} \left[\tilde{H}_r(P) \right] & \text{if } r \text{ is odd.} \end{cases}$$

We note that the finite set partition lattices satisfy the hypotheses of Proposition 5.34 since they are a special type of lattice called a geometric lattice. In [24], Folkman proved that geometric lattices of rank r have nonzero homology only in dimension r-2. Since Π_n is a geometric lattice of rank n-1, it has nonzero homology only in dimension n-3.

We are now ready to compute the (twisted) homology representations of intervals in the partition lattice in terms of the twisted homology of upper intervals.

⁽¹⁴⁾There is a very minor typo in Proposition 2.3 of [58] in what dimension the homology should be concentrated in.

LEMMA 5.35. Fix a multiset partition $S = \{\nu^{m_{\nu}} : \nu \in S\} \in \lambda(\mu)$ and let $r = \ell(\mu) - \ell(\lambda)$. As representations of $\prod_{\nu \in S} S_{m_{\nu}}[\operatorname{Stab}_{S|_{\nu|}}(X_{\nu})],$

$$\tilde{H}_{r-2}(X_S, Y_S) \otimes \det(X_S) \otimes \det(Y_S) \cong \bigotimes_{\nu \in S} \mathbb{1}_{S_{m_{\nu}}} \left[\tilde{H}_{\ell(\nu)-3}(X_{\nu}, \hat{1}) \otimes \det(X_{\nu}) \right].$$

Proof. The interval $[Y_S, X_S]$ in Π_n is (equivariantly) isomorphic to the following product of partition lattices:

$$\prod_{\nu \in S} \left([X_{\nu}, \hat{1}] \right)^{m_{\nu}},$$

with each $S_{m_{\nu}}[\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ acting on its corresponding $([X_{\nu}, \hat{1}])^{m_{\nu}}$. Recall the definition of $\rho(\nu)$ from Lemma 5.33. By Proposition 5.34, as $S_{m_{\nu}}[\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})]$ representations,

$$\tilde{H}_{r-2}\left(X_S, Y_S\right) \cong \bigotimes_{\nu \in S} \rho(\nu) \left[\tilde{H}_{\ell(\nu)-3}\left(X_{\nu}, \hat{1}\right)\right].$$

Now, by applying Lemma 5.33 and Lemma 5.13, we have the following isomorphism of $\prod_{\nu \in S} S_{m_{\nu}}[\operatorname{Stab}_{S|\nu|}(X_{\nu})]$ representations:

$$\tilde{H}_{r-2}(X_S, Y_S) \otimes \det(X_S) \otimes \det(Y_S) \cong \bigotimes_{\nu \in S} \mathbb{1}_{S_{m_{\nu}}} \left[\tilde{H}_{\ell(\nu)-3}(X_{\nu}, \hat{1}) \otimes \det(X_{\nu}) \right].$$

5.5. FINAL STEPS. In this section, we complete Step 4 of the proof outline in Section 5.1, proving our main theorem. To do so, we must first state a lemma and proposition. The following lemma appears as Exercise 4.1.3 in [28].

LEMMA 5.36. Let G_1 , G_2 be finite groups, with subgroups H_1 , H_2 , respectively. Let V_1 , V_2 be representations of H_1 , H_2 , respectively. Then, as $G_1 \times G_2$ representations,

$$(V_1 \otimes V_2) \Big|_{H_1 \times H_2}^{G_1 \times G_2} \cong V_1 \Big|_{H_1}^{G_1} \otimes V_2 \Big|_{H_2}^{G_2}.$$

PROPOSITION 5.37. Let λ and μ be any partitions of n. Write $m_{\nu}(S)$ to mean the multiplicity of a partition ν in a multiset partition S.

$$\operatorname{ch}(E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu}) = \sum_{S \in \lambda(\mu)} \prod_{\nu \in S} h_{m_{\nu}(S)} \left[\sum_{\substack{Lyndon \\ w}} \sum_{\substack{m \ge 1: \\ w^{m} \sim \nu}} L_{m}[h_{w}] \right].$$

Proof. If μ does not refine λ , then by Proposition 5.2, $E_{\lambda} \mathbb{C} \mathcal{F}_n E_{\mu} = 0$. On the right side, we also have 0, since $\lambda(\mu)$ is empty by definition.

Now, assume μ refines λ with $\ell(\mu) - \ell(\lambda) = r$. By Corollary 5.9, as S_n -representations,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{\substack{S_{n} - \text{orbits } [X \leqslant Y]:\\ X \in \mu \\ Y \in \lambda}} \tilde{H}_{r-2}(X,Y) \otimes \det(X) \otimes \det(Y) \uparrow_{\operatorname{Stab}_{S_{n}}(X) \cap \operatorname{Stab}_{S_{n}}(Y)}^{S_{n}}$$

Using Lemma 5.28 and Lemma 5.32,

$$E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu} \cong \bigoplus_{S \in \lambda(\mu)} \left(\tilde{H}_{r-2}(X_{S}, Y_{S}) \otimes \det(X_{S}) \otimes \det(Y_{S}) \right) \left(\prod_{\nu \in S} S_{m_{\nu}(S)} \left[\operatorname{Stab}_{S|_{\nu|}}(X_{\nu}) \right] \right) \right)$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

Using Lemma 5.35, the above is isomorphic (as S_n -representations) to

(23)
$$\bigoplus_{S\in\lambda(\mu)} \left(\bigotimes_{\nu\in S} \mathbb{1}_{S_{m_{\nu}(S)}} \left[\tilde{H}_{\ell(\nu)-3} \left(X_{\nu}, \hat{1} \right) \otimes \det \left(X_{\nu} \right) \right] \right) \Big|_{\prod_{\nu} S_{m_{\nu}(S)} \left[\operatorname{Stab}_{S_{|\nu|}} \left(X_{\nu} \right) \right]}^{S_{n}}.$$

We fix a particular $S = \{\nu^{m_{\nu}} : \text{partition } \nu\} \in \lambda(\mu)$, and manipulate the summand indexed by S as follows. By transitivity of induction and Lemma 5.36, the summand indexed by S above is isomorphic as an S_n -module to

$$\left(\bigotimes_{\nu\in S}\mathbb{1}_{S_{m_{\nu}}}\left[\tilde{H}_{\ell(\nu)-3}\left(X_{\nu},\hat{1}\right)\otimes\det\left(X_{\nu}\right)\right]^{S_{m_{\nu}}\cdot|\nu|}_{S_{m_{\nu}}\left[\operatorname{Stab}_{S_{|\nu|}}\left(X_{\nu}\right)\right]}\right)^{S_{n}}_{\Pi_{\nu}S_{m_{\nu}\cdot|\nu|}}\cong\\ \left(\bigotimes_{\nu\in S}\mathbb{1}_{S_{m_{\nu}}}\left[\tilde{H}_{\ell(\nu)-3}\left(X_{\nu},\hat{1}\right)\otimes\det\left(X_{\nu}\right)\right]^{S_{m_{\nu}}\left[S_{|\nu|}\right]}_{S_{m_{\nu}}\left[\operatorname{Stab}_{S_{|\nu|}}\left(X_{\nu}\right)\right]}\right]^{S_{m_{\nu}\cdot|\nu|}}_{S_{m_{\nu}}\left[\operatorname{Stab}_{S_{|\nu|}}\left(X_{\nu}\right)\right]}\right)^{S_{n}}_{\Pi_{\nu}S_{m_{\nu}\cdot|\nu|}}.$$

By Lemma 4.3, this is equivalent to

$$\bigotimes_{\nu \in S} \left(\mathbb{1}_{S_{m_{\nu}}} \left[\tilde{H}_{\ell(\nu)-3} \left(X_{\nu}, \hat{1} \right) \otimes \det \left(X_{\nu} \right) \uparrow_{\operatorname{Stab}_{S_{|\nu|}}(X_{\nu})}^{S_{|\nu|}} \right] \uparrow_{S_{m_{\nu}}[S_{|\nu|}]}^{S_{m_{\nu}\cdot|\nu|}} \right) \uparrow_{\Pi_{\nu}S_{m_{\nu}\cdot|\nu|}}^{S_{n}}.$$

However, by setting $\lambda = |\nu|, \mu = \nu$ in Corollary 5.9, this is the same as

$$\bigotimes_{\nu \in S} \left(\mathbb{1}_{S_{m_{\nu}}} \left[E_{|\nu|} \mathbb{C} \mathcal{F}_{|\nu|} E_{\nu} \right] \Big|_{S_{m_{\nu}}}^{S_{m_{\nu} \cdot |\nu|}} S_{m_{\nu}}^{S_{n}} \right) \Big|_{\Pi_{\nu} S_{m_{\nu} \cdot |\nu|}}^{S_{n}},$$

which has Frobenius characteristic $\prod_{\nu \in S} h_{m_{\nu}} \left[\operatorname{ch}(E_{|\nu|} \mathbb{C} \mathcal{F}_{|\nu|} E_{\nu}) \right]$. Returning to Eq. (23) and summing over all $S \in \lambda(\mu)$ reveals that

$$\operatorname{ch}(E_{\lambda}\mathbb{C}\mathcal{F}_{n}E_{\mu}) = \sum_{S\in\lambda(\mu)}\prod_{\nu\in S}h_{m_{\nu}(S)}\left[\operatorname{ch}(E_{|\nu|}\mathbb{C}\mathcal{F}_{|\nu|}E_{\nu})\right].$$

The proposition then follows from Proposition 5.1.

This next lemma follows from repeatedly applying Lemma 1.5 of [59].

LEMMA 5.38. Let f_1, f_2, \ldots, f_k be symmetric functions and t an abstract variable. Then,

$$\sum_{i \ge 0} t^i h_i [f_1 + f_2 + \ldots + f_k] = \prod_{j=1}^k \sum_{\ell \ge 0} t^\ell h_\ell [f_j].$$

We are finally ready to prove our main theorem.

THEOREM 5.39. There is an equality of generating functions

(24)
$$\sum_{n \ge 0} \sum_{\substack{\lambda \vdash n \\ \mu \vdash n}} \mathbf{y}_{\lambda} \mathbf{z}_{\mu} \cdot \operatorname{ch}(E_{\lambda} \mathbb{C} \mathcal{F}_{n} E_{\mu}) = \prod_{\substack{Lyndon \\ w}} \sum_{\substack{partition \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_{w}].$$

Algebraic Combinatorics, Vol. 8 #2 (2025)

Proof. By Proposition 5.37, it suffices to prove that

(25)
$$\sum_{n \ge 0} \sum_{\substack{\lambda \vdash n \\ \mu \vdash n}} \mathbf{y}_{\lambda} \mathbf{z}_{\mu} \cdot \sum_{S \in \lambda(\mu)} \prod_{\nu \in S} h_{m_{\nu}(S)} \left[\sum_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{m \ge 1: \\ w^m \sim \nu}} L_m[h_w] \right] \\= \prod_{\substack{\text{Lyndon partition} \\ w}} \sum_{\substack{\rho \cdot |w| \\ \rho}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_w]$$

We start by rewriting the right side of (25). Forming a partition ρ is the same as choosing a multiplicity $i \ge 0$ for each part size m > 0. Since $L_{1^{i_1}2^{i_2}...} = \prod_{m\ge 1} h_{i_m}[L_m]$ (Eq. (8)),

$$\prod_{\substack{\text{Lyndon partition}\\w}} \sum_{\substack{\rho \cdot |w| \\ p}} \mathbf{y}_{\rho \cdot |w|} \mathbf{z}_{w^{|\rho|}} L_{\rho}[h_w] = \prod_{\substack{\text{Lyndon}\\w}} \prod_{m \ge 1} \sum_{i \ge 0} \mathbf{y}_{(m \cdot |w|)^i} \mathbf{z}_{(w^{im})} h_i[L_m][h_w].$$

Using associativity of plethysm (Lemma 2.10(1)) to rewrite $h_i[L_m][h_w] = h_i[L_m[h_w]]$ and grouping together pairs (m, w) for which w^m rearranges to the same partition ν (written $w^m \sim \nu$), we can rewrite the above as

$$\prod_{\substack{\nu \neq \emptyset \\ \nu \neq \emptyset}} \prod_{\substack{(w,m): \\ w \text{ Lyndon,} \\ m \geqslant 1, \\ w^m \sim \nu}} \sum_{i \geqslant 0} (\mathbf{y}_{|\nu|} \mathbf{z}_{\nu})^i h_i [L_m[h_w]].$$

Setting $t = \mathbf{y}_{|\nu|} \mathbf{z}_{\nu}$ and using the $L_m[h_w]$ for $m \ge 1$ with $w^m \sim \nu$ as the f_i 's in Lemma 5.38, we can rewrite the result as

$$\prod_{\substack{\text{partition}\\\nu\neq\varnothing}}\sum_{i\geqslant 0} (\mathbf{y}_{|\nu|}\mathbf{z}_{\nu})^i h_i \left[\sum_{\substack{\text{Lyndon}\\w}}\sum_{\substack{m\geqslant 1:\\w^m\sim\nu}} L_m[h_w]\right].$$

Choosing a multiplicity $i \ge 0$ for each nonempty partition ν is the same as building a multiset partition S (with block ν appearing $i = m_{\nu}(S)$ times). Thus, we can rewrite the above as

$$\sum_{S} \prod_{\nu \in S} (\mathbf{y}_{|\nu|} \mathbf{z}_{\nu})^{m_{\nu}(S)} h_{m_{\nu}(S)} \left[\sum_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{m \geqslant 1: \\ w^m \sim \nu}} L_m[h_w] \right].$$

Г

٦

A multiset partition S lies in $\lambda(\mu)$ if and only if $\prod_{\nu \in S} (\mathbf{y}_{|\nu|} \mathbf{z}_{\nu})^{m_{\nu}(S)} = \mathbf{y}_{\lambda} \mathbf{z}_{\mu}$. Thus, we can rewrite the above to the desired generating function:

$$\sum_{n \geqslant 0} \sum_{\substack{\lambda \vdash n \\ \mu \vdash n}} \mathbf{y}_{\lambda} \mathbf{z}_{\mu} \cdot \sum_{S \in \lambda(\mu)} \prod_{\nu \in S} h_{m_{\nu}(S)} \left[\sum_{\substack{\text{Lyndon} \\ w}} \sum_{\substack{m \geqslant 1: \\ w^m \sim \nu}} L_m[h_w] \right].$$

6. Generalizations

6.1. OTHER COXETER TYPES. Some, but not all, of this work holds in general Coxeter types. In this section, we (briefly) explain partial generalizations and point out obstacles to full generalizations. For brevity, we assume the reader is familiar with reflection group theory and the face algebra of a reflection arrangement. See [32] for a

reference on reflection groups and [2, 48, 49] for nice descriptions of the general face and descent algebras.

Let W be a finite real reflection group with a set S of simple reflections. Write $\mathcal{F}(W)$ for the face monoid of the associated reflection arrangement and $\mathbb{CF}(W)$ for the face monoid algebra. Let $\Sigma(W)$ be W's associated descent algebra over the complex numbers. Bidigare proved his theorem [14, Theorem 3.8.1] at the level of general reflection groups, so there is an antiisomorphism $\Phi : (\mathbb{CF}(W))^W \to \Sigma(W)$.

Write $\mathcal{L}(W)$ for the lattice of intersections of the reflection arrangement of Wordered by *reverse* inclusion. The reflection group W acts on $\mathcal{L}(W)$ by poset automorphisms. There is again a support map $\sigma : \mathcal{F}(W) \to \mathcal{L}(W)$ which has the same nice properties as in type A (it commutes with the action of W, $\sigma(fg) = \sigma(f) \wedge \sigma(g)$, etc). The *pointwise stabilizer subgroup* of an intersection $X \in \mathcal{L}(W)$ is $W_X := \{w \in W : w(x) = x \text{ for all } x \in X\}$ and turns out to always be conjugate to a parabolic subgroup. (Recall that a parabolic subgroup is some subgroup W_J of Wgenerated by some subset $J \subseteq S$.)

The simple modules for Solomon's descent algebra are indexed by *W*-orbits of intersections [X] for $X \in \mathcal{L}(W)$. We write $X \in [X']$ if X is in the *W*-orbit of X' and write $\mathcal{L}(W)/W$ for the set of *W*-orbits of intersections. Similarly, we write $\mathcal{F}(W)/W$ for the set of *W*-orbits [F] of faces $F \in \mathcal{F}(W)$. Let $\{E_{[X]} : [X] \in \mathcal{L}(W)/W\}$ be a cfpoi for $(\mathbb{C}\mathcal{F}(W))^W$ and let $\{M_{[X]} : [X] \in \mathcal{L}(W)W\}$ denote the corresponding $\Sigma(W)$ -simples.

The analogue of Proposition 3.3 in general Coxeter type is the following proposition.

PROPOSITION 6.1. Let χ be an irreducible character of W. The χ -isotypic subspace of the $\mathbb{CF}(W)$ decomposes into $\Sigma(W)$ -modules as

$$(\mathbb{C}\mathcal{F}(W))^{\chi} = \bigoplus_{[X] \in \mathcal{L}(W)/W} \left(\mathbb{C}\mathcal{F}(W)E_{[X]}\right)^{\chi},$$

The \mathbb{C} -dimension of the submodule $(\mathbb{C}\mathcal{F}(W)E_{[X]})^{\chi}$ is

$$\dim(\chi) \cdot \#\{[F] \in \mathcal{F}(W)/W : \sigma(F) \in [X]\} \cdot \langle \chi, \mathbb{C}[W/W_X] \rangle.$$

Proof outline. The main ideas are:

- Direct analogues of Lemma 2.5, Lemma 2.6, Lemma 2.7, and Lemma 2.8 all hold because Aguiar–Mahajan [2] and Saliola [48] proved their results on cfpois for general finite reflection groups.
- The direct analogue to Lemma 3.2 also holds, giving a W-isomorphism between $\mathbb{CF}(W)E_{[X]}$ and the \mathbb{C} -span of faces $f \in \mathcal{F}(W)$ with $\sigma(F) \in [X]$. Since $\{w : w(F) = F\} = W_{\sigma(F)}$ and W_X is conjugate to $W_{X'}$ for $X' \in [X]$, the latter is W-isomorphic to the sum of coset representations

$$\bigoplus_{\substack{[F]\in\mathcal{F}(W):\\\sigma(F)\in[X]}} \mathbb{C}[W/W_{\sigma(F)}] \cong \bigoplus_{\substack{[F]\in\mathcal{F}(W):\\\sigma(F)\in[X]}} \mathbb{C}[W/W_X].$$

It would be nice if we could understand the composition multiplicities of the $\Sigma(W)$ simples $M_{[Y]}$ in each submodule $(\mathbb{CF}(W)E_{[X]})^{\chi}$. As in type A, we are able to reduce this to a question in (twisted) equivariant poset cohomology. There is a well-defined partial ordering on the W-orbits $[X] \in \mathcal{L}(W)/W$ given by declaring $[X] \leq [Y]$ if there exists $X' \in [X]$ and $Y' \in [Y]$ such that $X' \leq Y'$ in $\mathcal{L}(W)$. For an intersection X, let $N_W(X)$ denote the setwise W-stabilizer of X, i.e. $N_W(X) \coloneqq \{w \in W : w(X) = X\}$. (This notation is often used because $N_W(X)$ turns out to be the normalizer of W_X in W.) An analogue of Corollary 5.9 is below.

PROPOSITION 6.2. If $[X] \not\leq [Y]$, then $\left[\left(\mathbb{CF}(W) E_{[X]} \right)^{\chi} : M_{[Y]} \right] = 0$. Otherwise, if $k = \dim_{\mathbb{R}}(X) - \dim_{\mathbb{R}}(Y)$, then the composition multiplicity is

$$\left[\left(\mathbb{C}\mathcal{F}(W)E_{[X]} \right)^{\chi} : M_{[Y]} \right] = \dim \left(\chi \right) \cdot \left\langle \chi, E_{[Y]} \mathbb{C}\mathcal{F}(W)E_{[X]} \right\rangle.$$

Further, as W-modules, $E_{[Y]} \mathbb{CF}(W) E_{[X]}$ is isomorphic to

$$\bigoplus_{[X' \leq Y']} \left(\tilde{H}_{k-2}\left(X',Y'\right) \otimes \det(X') \otimes \det(Y') \right) \Big|_{N_W(X') \cap N_W(Y')}^W$$

The sum is over W-orbits of pairs $[X' \leq Y']$ in $\mathcal{L}(W)$ where $X' \in [X], Y' \in [Y]$.

Proof outline. As in type A (Proposition 3.5), finite dimensional algebra tools reveal the composition multiplicity

$$\left[\left(\mathbb{C}\mathcal{F}(W)E_{[X]} \right)^{\chi} : M_{[Y]} \right] = \dim\left(\chi \right) \cdot \left\langle \chi, E_{[Y]}\mathbb{C}\mathcal{F}(W)E_{[X]} \right\rangle.$$

Given that the analogues of Lemma 2.5 and Lemma 2.7 hold, we have the analogue of Proposition 5.2. Additionally, Saliola's equivariant surjection from the path algebra of the quiver to the face algebra described in Section 5.2.3 was a uniform result, which allows us to reframe $E_Y \mathbb{CF}_n E_X$ in terms of (twisted) poset cohomology by the argument in Section 5.2.5. Finally, the representations of real reflection groups are all self-dual as they are defined over the real numbers (see [8]), so the same tricks hold for converting cohomology to homology in the general case.

We are also able to understand the sign isotypic subspace for general type, gaining an analogue to Proposition 4.6. To state it, we first explain a known generalization of "cycle type" of an element $w \in W$. Let V be the ambient space of the reflection arrangement of W (i.e. the intersection at the bottom of $\mathcal{L}(W)$). Given an element $w \in W$, the space $\operatorname{Fix}(w) \coloneqq \{v \in V : w(v) = v\}$ turns out to be an intersection in $\mathcal{L}(W)$ ([40, Theorem 6.27]). Let the **cycle type** of w, $\operatorname{cyc}(w)$, be the W-orbit [Fix(w)].

In [15, Thm 1.1], Blessenohl, Hohlweg, and Schocker beautifully generalize Gessel and Reutenauer's result ([26, Theorem 2.1]) using connections between descent algebras and characters of reflection groups. One special case of their theorem proves $\langle \mathbb{C}W\Phi^{-1}(E_{[Y]}), \chi_{\text{sgn}} \rangle$ is the number of elements $w \in W$ with cycle type [Y] and descent set S.

PROPOSITION 6.3. As $\Sigma(W)$ -modules, the sign isotypic subspace of the face algebra $\mathbb{CF}(W)$ is isomorphic to the $\Sigma(W)$ -simple $M_{\text{cyc}(w_0)}$, where w_0 is the longest element of W.

Proof idea. It can be shown from Proposition 6.1 that $(\mathbb{CF}(W))^{\text{sgn}}$ is one dimensional and equal to $(\mathbb{CF}(W)E_{[V]})^{\text{sgn}}$. Hence it is isomorphic to the simple $M_{[Y]}$ for the unique [Y] for which

$$0 \neq \dim \left(E_{[Y]} \mathbb{C} \mathcal{F}(W) E_{[V]} \right)^{\operatorname{sgn}}$$

Using the same argument as in Section 4.4.2, one can show $E_{[Y]} \mathbb{CF}(W) E_{[V]}$ is isomorphic as a *W*-representation to $\mathbb{CW}\Phi^{-1}(E_{[Y]})$. Hence,

$$\dim \left(E_{[Y]} \mathbb{C}\mathcal{F}(W) E_{[V]} \right)^{\operatorname{sgn}} = \dim \left(\mathbb{C}W \Phi^{-1} \left(E_{[Y]} \right) \right)^{\operatorname{sgn}} = \left\langle \mathbb{C}W \Phi^{-1} \left(E_{[Y]} \right), \chi_{\operatorname{sgn}} \right\rangle.$$

The proposition then follows from Blessenohl–Hohlweg–Schocker's result.

Unfortunately, Proposition 6.1 and Proposition 6.3 are as far as we are able to generalize our work *uniformly* to general reflection groups. Indeed, the proof of our main theorem in type A (Theorem 5.39) relied heavily on symmetric function theory and the equivariant description of the homology of the lattice of intersections. As far as we are aware, there are no uniform generalizations of either of these notions. However,

from at least a naive standpoint, there appears to be more hope in generalizing this story specifically to type B.

Since the hyperoctahedral groups are wreath products of symmetric groups, there is a theory of type B symmetric functions: see [37, §5 of Appendix B] and also [42, §2]. The representation structure of the hyperoctahedral groups acting on the top (co)homology of the type B intersection lattice has also been studied: see [11, 27, 30]. Perhaps the most promising sign is that the type B descent algebra Cartan invariants are known (unlike in general Coxeter type). They were computed by N. Bergeron in [12], and are very closely connected to Garsia–Reutenauer's result in type A (Theorem 4.1). Hence, we pose the following question.

QUESTION 6.4. What is the type B analogue to Theorem 5.39?

Some useful first steps towards answering Question 6.4 would be to understand:

- the equivariant structure of upper intervals in the type B lattice of intersections (as Sundaram did for type A in Corollary 5.14),
- type *B* analogues of Lemma 5.18 and Proposition 5.1 (this may involve developing a hyperoctahedral analogue of Wachs's lemma (Lemma 5.17) and applying Poirier's work in [42, Theorem 16], which interprets a wreath product analog of L_{λ} as a necklace generating function), and
- the intersections of the setwise stabilizers of intersections, $N_W(X) \cap N_W(Y)$, and the *W*-orbits $[X \leq Y]$ (i.e. type *B* analogues of Lemma 5.28, Lemma 5.32).

6.2. LEFT REGULAR BANDS. A left regular band (or LRB) is a finite semigroup B for which $x^2 = x$ and xyx = xy for all $x, y \in B$. As discussed in the background (Eq. (1)), the face monoid of the braid arrangement-and in fact the face semigroups for all hyperplane arrangements-are examples of LRBs. LRBs have been studied extensively in [1, 20, 38, 39, 47, 57] and have fascinating connections with other mathematical areas such as Markov chains and poset topology. Many of the LRBs occurring in the literature come equipped with a natural group action and it is natural to consider the invariant theory of their semigroup rings. For instance, see [19] for a study of the invariant theory of the free LRB and its q-analog, which involved Stirling numbers, random-to-top shuffling, and the derangement symmetric function. Generalizations of the stories in [19] and this paper to general LRBs is ongoing thesis work of the author.

Acknowledgements. The author would like to thank her advisor Vic Reiner for proposing this project and providing truly invaluable guidance at every stage; Sheila Sundaram for several very helpful discussions about her work in [59] and for sparking the idea of an improved proof for step two of the proof outline; Franco Saliola for very helpful discussions early on about his work in [48, 49]; Esther Banaian, Sarah Brauner, Darij Grinberg, Trevor Karn, and Ben Steinberg for valuable references and conversations. Finally, the author wishes to thank the SageMath developers [61]. The author is supported by an NSF GRFP Award # 2237827 and also benefited from NSF grant DMS-2053288 for related travel.

References

- Marcelo Aguiar and Swapneel Mahajan, Coxeter groups and Hopf algebras, Fields Institute Monographs, vol. 23, American Mathematical Society, Providence, RI, 2006.
- [2] Marcelo Aguiar and Swapneel Mahajan, Topics in hyperplane arrangements, Mathematical Surveys and Monographs, vol. 226, American Mathematical Society, Providence, RI, 2017.
- [3] Connor Ahlbach and Joshua P. Swanson, Cyclic sieving, necklaces, and branching rules related to Thrall's problem, Electron. J. Combin. 25 (2018), no. 4, article no. 4.42 (38 pages).

- [4] Per Alexandersson, The symmetric functions catalog, https://www.symmetricfunctions.com.
- [5] Ibrahim Assem, Daniel Simson, and Andrzej Skowroński, Elements of the representation theory of associative algebras. Vol. 1. Techniques of representation theory, London Mathematical Society Student Texts, vol. 65, Cambridge University Press, Cambridge, 2006.
- [6] M. D. Atkinson, Solomon's descent algebra revisited, Bull. London Math. Soc. 24 (1992), no. 6, 545–551.
- [7] Hélène Barcelo, On the action of the symmetric group on the free Lie algebra and the partition lattice, J. Combin. Theory Ser. A 55 (1990), no. 1, 93–129.
- [8] Mark Benard, Schur indices and splitting fields of the unitary reflection groups, J. Algebra 38 (1976), no. 2, 318–342.
- [9] F. Bergeron and N. Bergeron, A decomposition of the descent algebra of the hyperoctahedral group. I, J. Algebra 148 (1992), no. 1, 86–97.
- [10] F. Bergeron, N. Bergeron, R. B. Howlett, and D. E. Taylor, A decomposition of the descent algebra of a finite Coxeter group, J. Algebraic Combin. 1 (1992), no. 1, 23–44.
- [11] Nantel Bergeron, A hyperoctahedral analogue of the free Lie algebra, J. Combin. Theory Ser. A 58 (1991), no. 2, 256–278.
- [12] Nantel Bergeron, A decomposition of the descent algebra of the hyperoctahedral group. II, J. Algebra 148 (1992), no. 1, 98–122.
- [13] Pat Bidigare, Phil Hanlon, and Dan Rockmore, A combinatorial description of the spectrum for the Tsetlin library and its generalization to hyperplane arrangements, Duke Math. J. 99 (1999), no. 1, 135–174.
- [14] Thomas Patrick Bidigare, Hyperplane arrangement face algebras and their associated Markov chains, Ph.D. thesis, University of Michigan, 1997.
- [15] Dieter Blessenohl, Christophe Hohlweg, and Manfred Schocker, A symmetry of the descent algebra of a finite Coxeter group, Adv. Math. 193 (2005), no. 2, 416–437.
- [16] Dieter Blessenohl and Hartmut Laue, The module structure of Solomon's descent algebra, J. Aust. Math. Soc. 72 (2002), no. 3, 317–333.
- [17] C. Bonnafé and G. Pfeiffer, Around Solomon's descent algebras, Algebr. Represent. Theory 11 (2008), no. 6, 577–602.
- [18] Sarah Brauner, Eulerian representations for real reflection groups, J. Lond. Math. Soc. (2) 105 (2022), no. 1, 412–444.
- [19] Sarah Brauner, Patricia Commins, and Victor Reiner, Invariant theory for the free left-regular band and a q-analogue, Pacific J. Math. 322 (2023), no. 2, 251–280.
- [20] Kenneth S. Brown, Semigroups, rings, and Markov chains, J. Theoret. Probab. 13 (2000), no. 3, 871–938.
- [21] Kenneth S. Brown and Persi Diaconis, Random walks and hyperplane arrangements, Ann. Probab. 26 (1998), no. 4, 1813–1854.
- [22] A. B. Dieker and F. V. Saliola, Spectral analysis of random-to-random Markov chains, Adv. Math. 323 (2018), 427–485.
- [23] Pavel Etingof, Oleg Golberg, Sebastian Hensel, Tiankai Liu, Alex Schwendner, Dmitry Vaintrob, and Elena Yudovina, *Introduction to representation theory*, Student Mathematical Library, vol. 59, American Mathematical Society, Providence, RI, 2011.
- [24] Jon Folkman, The homology groups of a lattice, J. Math. Mech. 15 (1966), 631–636.
- [25] A. M. Garsia and C. Reutenauer, A decomposition of Solomon's descent algebra, Adv. Math. 77 (1989), no. 2, 189–262.
- [26] Ira M. Gessel and Christophe Reutenauer, Counting permutations with given cycle structure and descent set, J. Combin. Theory Ser. A 64 (1993), no. 2, 189–215.
- [27] Eric Gottlieb and Michelle L. Wachs, Cohomology of Dowling lattices and Lie (super)algebras, Adv. in Appl. Math. 24 (2000), no. 4, 301–336.
- [28] Darij Grinberg and Victor Reiner, Hopf algebras in combinatorics, 2014, https://arxiv.org/ abs/1409.8356.
- [29] Phil Hanlon, The fixed-point partition lattices, Pacific J. Math. 96 (1981), no. 2, 319-341.
- [30] Phil Hanlon, The characters of the wreath product group acting on the homology groups of the Dowling lattices, J. Algebra 91 (1984), no. 2, 430–463.
- [31] A. Jöllenbeck and C. Reutenauer, Eine Symmetrieeigenschaft von Solomons Algebra und der höheren Lie-Charaktere, Abh. Math. Sem. Univ. Hamburg 71 (2001), 105–111.
- [32] Richard Kane, Reflection groups and invariant theory, CMS Books in Mathematics/Ouvrages de Mathématiques de la SMC, vol. 5, Springer-Verlag, New York, 2001.
- [33] Aleksandr Anatol'evich Klyachko, Lie elements in a tensor algebra, Sibirsk. Mat. Ž. 15 (1974), 1296–1304, 1430.

- [34] Nadia Lafrenière, Eigenvalues of symmetrized shuffling operators, Sém. Lothar. Combin. 82B (2020), article no. 78 (12 pages).
- [35] Nicholas A. Loehr and Jeffrey B. Remmel, A computational and combinatorial exposé of plethystic calculus, J. Algebraic Combin. 33 (2011), no. 2, 163–198.
- [36] M. Lothaire, Combinatorics on words, Cambridge Mathematical Library, Cambridge University Press, Cambridge, 1997.
- [37] I. G. Macdonald, Symmetric functions and Hall polynomials, second ed., Oxford Classic Texts in the Physical Sciences, The Clarendon Press, Oxford University Press, New York, 2015.
- [38] Stuart Margolis, Franco Saliola, and Benjamin Steinberg, Combinatorial topology and the global dimension of algebras arising in combinatorics, J. Eur. Math. Soc. (JEMS) 17 (2015), no. 12, 3037–3080.
- [39] Stuart Margolis, Franco V. Saliola, and Benjamin Steinberg, Cell complexes, poset topology and the representation theory of algebras arising in algebraic combinatorics and discrete geometry, Mem. Amer. Math. Soc. 274 (2021), no. 1345, xi+135.
- [40] Peter Orlik and Hiroaki Terao, Arrangements of hyperplanes, Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], vol. 300, Springer-Verlag, Berlin, 1992.
- [41] Götz Pfeiffer, A quiver presentation for Solomon's descent algebra, Adv. Math. 220 (2009), no. 5, 1428–1465.
- [42] Stéphane Poirier, Cycle type and descent set in wreath products, Discrete Math. 180 (1998), no. 1-3, 315–343, Proceedings of the 7th Conference on Formal Power Series and Algebraic Combinatorics (Noisy-le-Grand, 1995).
- [43] Victor Reiner, Franco Saliola, and Volkmar Welker, Spectra of symmetrized shuffling operators, Mem. Amer. Math. Soc. 228 (2014), no. 1072, vi+109.
- [44] Christophe Reutenauer, Free Lie algebras, London Mathematical Society Monographs. New Series, vol. 7, The Clarendon Press, Oxford University Press, New York, 1993, Oxford Science Publications.
- [45] Jay-Calvin Uyemura Reyes, Random walk, semi-direct products, and card shuffling, Ph.D. thesis, Stanford University, 2002.
- [46] Bruce E. Sagan, The symmetric group: representations, combinatorial algorithms, and symmetric functions, second ed., Graduate Texts in Mathematics, vol. 203, Springer-Verlag, New York, 2001.
- [47] Franco V. Saliola, The quiver of the semigroup algebra of a left regular band, Internat. J. Algebra Comput. 17 (2007), no. 8, 1593–1610.
- [48] Franco V. Saliola, On the quiver of the descent algebra, J. Algebra **320** (2008), no. 11, 3866–3894.
- [49] Franco V. Saliola, The face semigroup algebra of a hyperplane arrangement, Canad. J. Math. 61 (2009), no. 4, 904–929.
- [50] Franco V. Saliola, The Loewy length of the descent algebra of type D, Algebr. Represent. Theory 13 (2010), no. 2, 243–254.
- [51] Manfred Schocker, Multiplicities of higher Lie characters, J. Aust. Math. Soc. 75 (2003), no. 1, 9–21.
- [52] Manfred Schocker, The descent algebra of the symmetric group, in Representations of finite dimensional algebras and related topics in Lie theory and geometry, Fields Inst. Commun., vol. 40, Amer. Math. Soc., Providence, RI, 2004, pp. 145–161.
- [53] Manfred Schocker, The module structure of the Solomon-Tits algebra of the symmetric group, J. Algebra 301 (2006), no. 2, 554–586.
- [54] Louis Solomon, A Mackey formula in the group ring of a Coxeter group, J. Algebra 41 (1976), no. 2, 255–264.
- [55] Richard P. Stanley, Some aspects of groups acting on finite posets, J. Combin. Theory Ser. A 32 (1982), no. 2, 132–161.
- [56] Richard P. Stanley, *Enumerative combinatorics. Vol. 2*, Cambridge Studies in Advanced Mathematics, vol. 62, Cambridge University Press, Cambridge, 1999.
- [57] Benjamin Steinberg, Representation theory of finite monoids, Universitext, Springer, Cham, 2016.
- [58] Sheila Sundaram, Applications of the Hopf trace formula to computing homology representations, in Jerusalem combinatorics '93, Contemp. Math., vol. 178, Amer. Math. Soc., Providence, RI, 1994, pp. 277–309.
- [59] Sheila Sundaram, The homology representations of the symmetric group on Cohen-Macaulay subposets of the partition lattice, Adv. Math. 104 (1994), no. 2, 225–296.
- [60] Sheila Sundaram, Plethysm, partitions with an even number of blocks and Euler numbers, in Formal power series and algebraic combinatorics (New Brunswick, NJ, 1994), DIMACS Ser.

Discrete Math. Theoret. Comput. Sci., vol. 24, Amer. Math. Soc., Providence, RI, 1996, pp. 171–198.

- [61] The Sage Developers, SageMath, the Sage Mathematics Software System (Version 10.1), 2024, https://www.sagemath.org.
- [62] R. M. Thrall, On symmetrized Kronecker powers and the structure of the free Lie ring, Amer. J. Math. 64 (1942), 371–388.
- [63] Jacques Tits, Two properties of Coxeter complexes, J. Algebra 41 (1976), no. 2, 265–268.
- [64] Michelle L. Wachs, Whitney homology of semipure shellable posets, J. Algebraic Combin. 9 (1999), no. 2, 173–207.
- [65] Michelle L. Wachs, Poset topology: tools and applications, in Geometric combinatorics, IAS/Park City Math. Ser., vol. 13, Amer. Math. Soc., Providence, RI, 2007, pp. 497–615.
- [66] Peter Webb, A course in finite group representation theory, Cambridge Studies in Advanced Mathematics, vol. 161, Cambridge University Press, Cambridge, 2016.

PATRICIA COMMINS, University of Minnesota, School of Mathematics, 206 Church St SE, Minneapolis, MN 55455 (USA)

 $E\text{-}mail:\texttt{commi010@umn.edu}\\ Url:\texttt{https://sites.google.com/umn.edu/patty-commins}$